

POWER NETWORK METER END25

USER'S MANUAL

1. INTRODUCTION

The Multifunction Instrument is a panel mounted 96 x 96mm DIN Quadratic Digital Panel Meter, which measures important electrical parameters in 3 ph 4 wire / 3 wire / 1 ph Network and replaces the multiple analog panel meters. It measures electrical parameters like AC voltage, Current, Frequency, Power, Energy (Active / Reactive / Apparent), phase angle, power factor, individual harmonics & many more. The instrument integrates accurate measurement technology (All Voltages & current measurements are True RMS upto 31st Harmonic) with LCD display with backlit.

It can be configured & Programmed at site for the following : PT Primary, PT Secondary, CT Primary, CT Secondary 3 Phase 3W, 3 Phase 4W, 1 Phase 2W system.

The front panel has three push buttons using which the user can scroll through different screens & configure the product. The front panel also has Impulse red led, flashing at rate proportional to measured power.



2. MEASUREMENT AND ENERGY/COUNTER SCREENS

In normal operation, the user is presented with two simultaneous screens

1. One of the Measurement screens out of the screens 1 to 36 of Table . These screens may be scrolled through one at a time in incremental order by pressing the "Up key" and in decremental order by pressing "Down key". Few important screens are explained in Section 2.1 to 2.6.**

2.One of the Energy/Counter screens out of the screens 37 to 62 of Table 1. These screens may be scrolled through one at a time in incremental order only by pressing the "Enter key" to roll over again in the same order. Few important screens are explained in Section 2.7 to 2.9**.

Load Graphics

Load Graphics indicates the input current as the percentage of the CT Primary value. This indication is available for all measurement screens. For example, consider CT Primary to be set at 5A, then the input current of 2.5 A indicates 50% as shown below.

** - See full version of service manual



The absence of lines indicating the percentage implies that the input current is less than 20% of the CT Primary Phase Sequence Indication

It indicates the rotation of input phasor vectors : clockwise/counter-clockwise.

In case the input is absent or the phase sequence is neither L123 nor L321, the phase sequence indication is not

Counter-Clockwise Sequence, for L321 Clockwise Sequence for L123.

TABLE 1 : Measurement & Energy/Counter Screens for Basic version of END25

Screen	Parameters		On Display			On Modbus			
No.			3P 3W	1P 2W	3P 4W	3P 3W	1P 2W		
1	System Voltage/ Current/ Power/ Frequency	√	\checkmark	√	√	√	√		
2	L-N Voltage	✓	ű	ű	√	ű	ű		
3	L-L Voltage	√	\checkmark	ű	\checkmark	√	ű		
4	Current ,Neutral Current	✓	√*	ű	√	√*	ű		
5	Phase Voltage THD	ű	ű	ű	ű	ű	ű		
6	Phase Current THD	ű	ű	ű	ű	ű	ű		
7	L1 phase VA/VAr/W	✓	ű	ű	✓	ű	ű		
8	L2 phase VA/VAr/W	√	ű	ű	√	ű	ű		
9	L3 phase VA/VAr/W	√	ű	ű	\checkmark	ű	ű		
10	Phase Angle	ű	ű	ű	ű	ű	ű		
11	VA/A demand	ű	ű	ű	ű	ű	ű		
12	Capacitive, Inductive VAr demand	ű	ű	ű	ű	ű	ű		
13	W IMP demand	ű	ű	ű	ű	ű	ű		
14	W EXP demand	ű	ű	ű	ű	ű	ű		
15	Max VA/A demand	ű	ű	ű	ű	ű	ű		
16	Max Capacitive, Inductive VAr demand	ű	ű	ű	ű	ű	ű		
17	Max W IMP demand	ű	ű	ű	ű	ű	ű		
18	Max W EXP demand	ű	ű	ű	ű	ű	ű		

Screen				On Display			On Modbus			
No.	Parameters	3P 4W	3P 3W	1P 2W	3P 4W	3P 3W	10.20			
19	Old Max VA/A demand			ű	<u> </u>					
20	Old Max Capacitive Inductive VAr demand	l ű	u ű	u ű	l ű	l ű	l ű			
21	Old Max W IMP demand	- u ű	ű	ű	l ű	l ű	l ű			
22	Old Max W EXP demand	- u ű	ű	ű	l ű	l ű	l ű			
23	Svs RPM/Frequency	1	<u> </u>	u V	4	1	1			
24	Svs Active/Reactive/Apparent power	1	1	1	1	1	1			
25	Svs Apparent, Reactive (Power), Phase angle, PF	1	1	1	1	1	1			
26	Min Sys Voltage / Current	√	√	√	√	\checkmark	√			
27	Max Sys Voltage / Current	1	1	1	1	\checkmark	1			
28	Sys %THD Voltage /Current	V	√	√	√	√	√			
29	Current Reversal	1	ű	\checkmark	ű	ű	ű			
30	Phase rotation error	V	V	ű	1V	1V	ű			
31	Phase absent indication	1	ű	ű	ű	ű	ű			
32	RTC	V	\checkmark	√	\checkmark	\checkmark	\checkmark			
33	Individual harmonics V	ű	ű	ű	ű	ű	ű			
34	Individual harmonics A	ű	ű	ű	ű	ű	ű			
35	Timer1 No. of Cycles, ON, OFF delay	V	√	 ✓ 	 ✓ 	\checkmark	\checkmark			
36	Timer2 No. of Cycles, ON, OFF delay	 ✓ 	√	√	 ✓ 	 ✓ 	 ✓ 			
37	Active Energy Import(Overflow)	ű	ű	ű	√	✓	\checkmark			
38	Active Energy Import	√	✓	✓	√	√	√			
39	Active Energy Export(Overflow)	ű	ű	ű	\checkmark	 ✓ 	\checkmark			
40	Active Energy Export	✓	✓	✓	✓	 ✓ 	 ✓ 			
41	Reactive Capacitive energy (Overflow)	ű	ű	ű	\checkmark	 ✓ 	\checkmark			
42	Reactive Capacitive energy	✓	\checkmark	\checkmark	\checkmark	\checkmark	√			
43	Reactive Inductive energy (Overflow)	ű	ű	ű	√	✓	✓			
44	Reactive Inductive energy	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	√			
45	Apparent energy(Overflow)	ű	ű	ű	✓	✓	√			
46	Apparent energy	√	\checkmark	\checkmark	\checkmark	 ✓ 	√			
47	Run hour	 ✓ 								
48	On hour	 ✓ 								
49	No. of interrupts	 ✓ 	\checkmark							
50	Old Active energy Import(Overflow)	ű	ű	ű	ű	ű	ű			
51	Old Active energy Import	ű	ű	ű	ű	ű	ű			
52	Old Active energy Export(Overflow)	ű	ű	ű	ű	ű	ű			
53	Old Active energy Export	ű	ű	ű	ű	ű	ű			
54	Old Reactive Capacitive energy (Overflow)	ű	ű	ű	ű	ű	ű			
55	Old Reactive Capacitive energy	ű	ű	ű	ű	ű	ű			
56	Old Reactive Inductive energy (Overflow)	ű	ű	ű	Ű	ű	<u>ű</u>			
57	Old Reactive Inductive energy	ű	ű	ű	ű	ű	Ű			
58	Old Apparent energy(Overflow)	ű	ű	ű	ű	ű	ű			
59	Old Apparent energy	ű	Ű	Ű	Ű	Ű	Ű			
60	Old Run hour	ű	Ű	ű	ű	ű	Ű			
61	Old On hour	ű	ű	l ű	<u> </u>	l ű	<u> </u>			
62	Old no. of interrupts	Ű	Ű	Ű	l ü	μű	ΙŰ			

Note: (1) Only screens (with screen number) 1 to 32 are available for selectable Userscreens (2) Energy on Display is Auto-Ranging For details refer Section 4.1**.

*Note: In 3P3W system, Neutral Current is not shown, only line currents are shown. ** - See full version of service manual.



** - See full version of service manual





3. PROGRAMMING

The following sections comprise step by step procedures for configuring the Multifunction Meter according to individual user requirements. To access the set-up screens press and hold " ▲ UP" and " ▼ DOWN" keys simultaneously for 5 seconds. This will take the User into the Password Protection Entry Stage (Section 3.1)**.

3.1. Password Protection

Password protection can be enabled to prevent unauthorised access to set-up screens, when default password protection is not enabled. Password protection is enabled by selecting a four digit number othe than 0000, setting a password of 0000 disables the password protection.

	Enter Password, prompt for first digitPress the "▲" key to scroll the value of first digit from 0 through to 9, the value rolls back from 9 round to 0 and " ▼ " key to scroll the value of first digit from 9 through to 0, the value rolls back from 0 round to 9. Press the "—" key to advance to next digit.
LodE DDDD donE	In special case where the Password is "0000" pressing the "_" key when prompted for the first digit advances to the password accepted screen and then pressing the "_" key again makes the set-up screens accessible to the user. But instead of pressing the "_" key, if " ▲ " or " ▼ " key is pressed, the user is taken to the "New/change Password" entry stage.
LodE {	Enter Password, first digit entered, prompt for second digit Press the "▲" key to scroll the value of first digit from 0 through to 9, the value rolls back from 9 round to 0 and " ▼ " key to scroll the value of first digit from 9 through to 0, the value rolls back from 0 round to 9. Press the "■" key to advance to next digit.
CodE (3	Enter Password, second digit entered, prompt for third digit. Press the "▲" key to scroll the value of first digit from 0 through to 9, the value rolls back from 9 round to 0 and "♥" key to scroll the value of first digit from 9 through to 0, the value rolls back from 0 round to 9. Press the "—" key to advance to next digit.
EodE 134-	Enter Password, third digit entered, prompt for fourth digit Press the "▲" key to scroll the value of first digit from 0 through to 9, the value rolls back from 9 round to 0 and " ♥ " key to scroll the value of first digit from 9 through to 0, the value rolls back from 0 round to 9. Press the "—" key to advance to verification of the password.
EodE 1342	Enter Password, fourth digit entered, awaiting verification of the password.

Password confirmed

EodE 1342 donE

Password Incorrect.

New / Change Password









Pressing " 🔺 " or " 🔻 " key advances to the "New / change Password" entry stage.

Pressing the "- " key advances to the Menu selection (setup menu) screen (see Section 3.2).

The unit has not accented the Password entered

Pressing the "▲ " or " ▼ " key advances to the Enter Password stage. Pressing the " - " key exits the Password menu & returns operation to the

measurement reading mode

Prompting for first digit.

Press the "▲" and " ▼ " keys to scroll the value of first digit from 0 through to 9 and from 9 through to 0, respectively with digit roll around feature

Pressing the "-" key advances the operation to the next digit and sets the first digit, in this case to "2".

New/ Change Password, first digit entered, prompting for second digit.

Press the " A " and " V " keys to scroll the value of second digit from 0 through to 9 and from 9 through to 0,respectively with digit roll around feature.

Pressing the "_ " key advances the operation to the next digit and sets the second digit, in this case to "1"

New/ Change Password, second digit entered, prompting for third digit.

Press the "▲" and "▼" keys to scroll the value of second digit from 0 through to 9 and from 9 through to 0,respectively with digit roll around feature.

Pressing the "- " key to advance the operation to the next digit and sets the third digit, in this case to "5".

New/ Change Password, third digit entered, prompting for fourth digit. .

Press the " \blacktriangle " and " \blacktriangledown " keys to scroll the value of second digit from 0 through to 9 and from 9 through to 0,respectively with digit roll around feature.

Pressing the "-" key to advance the "New Password Confirmed" and sets the fourth digit, in this case to "3".

Pressing the " **▲** " or " **▼** " key returns to the "New/Change Password"

Pressing the " \blacksquare " key advances to the Menu selection screen (see Section 3.2).

3.2 Menu selection. 3.2.1 System Parameter Selection



This screen is used to select different system Parameters like "system type","CT Ratio","PT Ratio", etc. Pressing the " — " key allows the user to set Different system parameters (see Section 3.2.1.1 to 3.2.1.12)**.

Pressing the " A " key advances to the "Communication Parameter Selection" screen (see section 3.2.2) and pressing " V " key advances to the "Quit Setup" Screen (see section 3.2.9).

3.2.2 Communication Parameter Selection



This screen is used to select the different communication parameters like "Address selection", "RS485 Parity selection", "RS485 baud rate", etc. Pressing the " ■ " key allows the user to set different Communication parameters (see Section 3.2.2.1 to 3.2.2.3)**. Pressing the "▲" key advances to the "Reset Parameter Selection" screen (if ethernet module is present then it will advance to IP parameter selection see section 3.2.3) (see section 3.2.4 and pressing " ♥ " key advances to the "System Parameter Selection" screen (see Section 3.2.1).

3.2.3 IP Parameter Selection



5EL

SELUP

r SEL PRr R

This screen is used to select the different communication parameters like "IP Address selection", "Subnet mask selection", "default gateway selection".server port selection etc.

Pressing the "- " key allows the user to set different Ethernet parameters (see Section 3.2.3.1 to 3.2.3.5)**

Pressing the "▲" key advances to the "Reset Parameter Selection" screen (see section 3.2.4) and pressing "▼" key advances to the "Communication Parameter Selection" screen (see Section 3.2.2.)

3.2.4 Reset Parameter Selection

This screen is used to reset different parameters.

Pressing the " — " key allows the user to reset different system parameters (see Section 3.2.4.1)**.

Pressing the *▲ * key advances to the "Output Option Selection" screen (see Section 3.2.5) and pressing *▼ * key advances to "Communication Parameter Selection" screen(ethermet module is present then it advances to IP parameter selection else it goes to serial communication parameter selection) (see Section 3.2.2).

** - See full version of service manual

3.2.5 Output Option Selection

This screen will allow the user to select Output Options like "Relay Output".

5EL SELuP

Pressing the " - " key allows the user to select and configure the output options (see Section 3.2.5.1)**. Pressing " \blacktriangle " key advances to the "Datalog Option Selection" screen (see section 3.2.6) and pressing " \checkmark " key advances to "Reset Parameter Selection" screen (see Section 3.2.4).

3.2.6 Datalog Option Selection

5EL <u> ABF</u>B LoG SELUP

This screen will allow the user to select Datalog Options like "Event Based", "Time Based" and "Load Profile Pressing the " — " key allows the user to select and configure the datalog options (see section 3.2.6.1)**.

Pressing the "▲" key advances to "Display Parameters" screen (see section 3.2.7) and pressing "♥" key advances to "Output Option Selection" screen (see Section 3.2.5).

3.2.7 Display Parameters



This screen will allow the user to access different features like "Backlit", "Screens", "Contrast" and "Old Parameters". Pressing the " - " key allows the user to select and configure the features (see Section 3.2.7.1)**.

Pressing the " \blacktriangle " key advances to "RTC Setting" screen (see Section 3.2.8) and pressing " \blacktriangledown " key advances to "Datalog Option Selection" screen (see Section 3.2.6).

3.2.8 RTC Setting



This screen will allow the user to access features like "Set Date" and "Set

Pressing the "— " key allows the user to select date and time (see Section 3.2.8.1)**.

Pressing the "▲" key advances to "Factory Reset" screen (see Section 3.2.9) and pressing "▼" key advances to "Display Parameters" screen (see Section 3.2.7).

3.2.9 Factory Reset



This screen allows the user to set the meter to its Factory Default settings (see Section 3.2.9.1)**.

Pressing the "▲" key advances to "Quit Setup"screen (see Section 3.2.10) and pressing " ▼ " key advances to "RTC Setting" screen (see Section and pressing " ▼ 3.2.9).

3.2.10 Quit Setup



This screen will allow the user to quit the setup menu (see Section 3.2.10.1)** Pressing the "- " key allows the user to Quit from setup menu & return to measurement screen

Easy Clip-in

mounting

1 - 6mm for swivel screw

Panel Thickness : 1 - 3mm for self clicking

Pressing the "▲" key advances to "System Parameter Selection" screen (see Section 3.2.1) and pressing "▼" key advances to "Factory Reset" screen (see Section 3.2.9).

** - See full version of service manual.

4. Installation

Mounting of the Meter is featured with easy "Clip-in" mounting. Push the meter in panel slot (size 92 x92 mm), it will click fit into panel with the four integral retention clips on two sides of meter. If required, additional support is provided with swivel screws as shown in figure.

The front of the enclosure conforms to IP54. Additional protection to the panel may be obtained by the use of an Optional panel gasket. The terminals at the rear of the product should be protected from liquids

The Meter should be mounted in a reasonably stable ambient temperature and where the operating temperature is within the range 0 to 50°C. Vibration should be kept to a minimum and the product should not be mounted where it will be subjected to excessive direct sunlight.

Caution

1. In the interest of safety and functionality this product must be installed by a qualified engineer, abiding by any local regulations.

2. Voltages dangerous to human life are present at some of the terminal connections of this unit. Ensure that all supplies are de-energised before attempting any connection or disconnection. 3. These products do not have internal fuses therefore external fuses must be used to ensure safety

4.1 EMC Installation Requirements

This product has been designed to meet the certification of the EU directives when installed to a good code of

In sproduct has been designed to meet the certification of the EU directives when installed to a good code of practice for EMC in industrial environments, e.g. 1. Screened output and low signal input leads or have provision for fitting RF suppression components, such as ferrite absorbers, line filters etc., in the event that RF fields cause problems. Note : It is good practice to install sensitive electronic instruments that are performing critical functions, in EMC enclosures that protect against electrical interference which could cause a disturbance in function. 2. Avoid routing leads alongside cables and products that are, or could be, a source of interference. To protect the product the protect against electrical interference when protect against electrical interference when protect against electrical interference when the protect against electrical interference interference.

2. Noto totaling leade alongside coulds and provide a single could be along of the product against permanent damage, surge transients must be limited to 2kV pk. It is good EMC practice to suppress differential surges to 2kV at the source. Theunit has been designed to automatically recover in the event of a high level of transients. In extreme circumstances it may be necessary to temporarily disconnect the auxiliary supply for a period of greater than 5 seconds to restore correct operation.

The Current inputs of these products are designed for connection in to systems via Current Transformers only, nere one side is arounder where one side is grounded.
 4. ESD precautions must be taken at all times when handling this product.

4.2 Case Dimensions and Panel Cut-Out



4.3 Wiring

Input connections are made directly to screw-type terminals with indirect wire pressure. Numbering is clearly marked on the connector. Choice of cable should meet local regulations. Terminal for both Current and Voltage inputs will accept upto 4mm² (12AWG) solid or 2.5 mm² stranded cable. ded to use wire with lug for connection with meter.

4.4 Auxiliary Supply

Meter should ideally be powered from a dedicated supply, however powered from the signal source, provided the source remains within it may be the limits of the Chosen auxiliary voltage range.

4.5 Fusing

ended that all voltage lines are fitted with 1 Amp HRC fuse It is recom

4.6 Earth/Ground Connections For safety reasons, CT secondary connections should be grounded in accordance with local regulations

5. Connection Diagrams







6. Pluggable Module



7. Specification

System

3 Phase 3 Wire / 4 Wire or Single Phase programmable on site

Inputs Nominal Input Voltage (AC RMS)

100VLL to 600VLL programmable on site (57.7VLN to 346.4VLN) 100VLL to 1200 kVLL, programmable on site System PT Primary Values 100VLL to 600VLL programmable on site. (57.7VLN to 346.4VLN)

System PT Secondary Values Max continuous input voltage

Nominal input voltage burden

Nominal Input Current

max continuous input current Nominal input current burden System CT primary values

System CT secondary values Overload Indication

Overload withstand Voltage input

Current input

2 x Rated Value (1s application at 10s intervals) repeated 10 times 2 x nominal Value (1s application at 5min intervals) repeated 5 times

<0.3 VA approx. per Phase (at nominal 240V)

Std. Values 1 to 9999A (1 or 5 Amp secondary)

Auxiliary Supply External Higher Aux. External Higher Aux. Nominal Value

100V to 550V AC-DC 230V AC/DC 50/60 Hz for AC Aux Aux Frequency Range 45 to 65 Hz < 6 VA approx. (at nominal) VA Burden With Addon card VA Burden With Ethernet card < 8 VA approx. (at nominal)

Operating Measuring Ranges 20 ... 120 % of nominal Value

Current Starting Current

Total Harmonic Distortion

Voltage

Frequency

Power Factor

45 to 66 Hz 0.5 Lag ... 1 ... 0.8 Lead 50% upto 15th harmonic

10% upto 31st harmonic

± 0.5% of nominal value

± 0.5% of nominal value

+ 0.2 % of mid frequency

± 0.5% of nominal value

± 1.0% of nominal value

± 0.5% of nominal value

as per IEC 62053-22 (0.5s)

Class 2 as per IEC 62053-23

as per IEC 62053-22 (0.2s) (optional)

±3°

class 1

± 3.0 %

23°C <u>+</u> 2°C

1....120% of nominal value

as per IEC 62053-22 (0.5s/0.2s)

120% of Nominal Value

120% of Nominal value

<0.3 VA approx. per phase

1A / 5A, programmable on site

"-OL-" >121% of Nominal value (for voltage and current)

14/54

Reference conditions for Accuracy Reference temperature

Accuracy Voltage Current Frequency Active power Reactive power

(as per IEC 62053-21)

Apparent Power Power Factor / Phase Angle Active energy

Reactive energy

Apparent energy

THD (Voltage / Current) Accuracy (for optional 0.2s)

Voltage Current Frequency Active power Apparent Power

Display LCD Display with backlit Response time to step input

Controls User Interface

Standards EMC Immunity Immunity

Safety

IP for water & dust

+ 0.2% of nominal value + 0.2% of nominal value + 0.2 % of mid frequency ± 0.2% of nominal value

+ 0.2% of nominal value

Approx. 1 sec.

3 push buttons

IEC 61326 - 1 : 2012 IEC 61000-4-3. 10V/m -Level 3 industrial Low level IFC 61010-1-2010 permanently connected use IEC 60529

Pollution degree	2
Installation Category	III
loolation	
Protective Class	2
High Voltage Test	-
Input+Aux vs Surface	4 kV RMS, 50Hz, 1 min
Input vs Remaining Circuit	3.3 kV RMS, 50Hz, 1 min
Environmental conditions	
Operating temperature	-10 to +60 °C
Storage temperature	-20 to +70 °C
Relative humidity	0 95 % RH (Non condensing)
Warm up time	3 minute (minimum)
Shock (As per	Half Sine wave, Peak acceleration
IEC 60068-2-27)	30gn (300 m/s^2), duration 18 ms
Vibration	1015010 Hz, 0.15mm amplitude
Number of Sweep cycles	10 per axis
Enclosure	
Enclosure front	IP 54
Enclosure back (Terminals)	IP 20
Dimensione	
Dimensions	
Bezel Size	96mm x 96mm DIN 43718
Panel cut out	92mm+0.8 X 92mm+0.8
Overall Depth	75 mm
Panel thickness	1 - 3mm for self clicking
Weight	1 - 6mm for swivel screws
weight	320 grams Approx.
Pulse output Option	
Relay	1NO
Switching Voltage & Current	240 VAC , 5 A.
Default Pulse rate Divisor	1 per Wh (up to 3600W),
	1 per kWh (up to 3600kW), 1 per MWh (above 3600 kW) up to 30000 kW)
Pulse rate Divisors	Programmable on site
10	1 per 10Wh (up to 3600W)
10	1 per 10kWh (up to 3600kW),
	1 per 10MWh (above 3600 kW up to30000 kW)
100	1 per 100Wh (up to 3600W), 1 per 100kW/h (up to 3600 kW)
	1 per 100MWh (above 3600 kW),
1000	1 per 1000\M/b (up to 2600\M/)
1000	1 per 1000kWh (up to 3600kW).
	1 per 1000MWh (above 3600 kW up to 30000 kW)
Pulse Duration	60ms , 100ms or 200ms
Note:	
2. Above conditions are also applica	ble for Reactive & Apparent Energy.
ModBus (RS 485) Option :	
Protocol	ModBus (RS 485)
Baud Rate	4.8k, 9.6k, 19.2k, 38.4k, 57.6kbps
Parity	Odd or Even, with 1 stop bit, Or None with 1 or 2 stop bits
Ethernet	Ethernet access on Modbus TCP/IP Protocol
Impulse Output :	For Energy testing
IIIpuise LED	For Energy testing

8. Connection for Optional Pulse Output / RS 485 (rear view of Multifunction Meter)

1. RS 485 Output with Relay1 & Relay2

	$\begin{bmatrix} r & r & r & r \\ 0 & 0 & 0 \end{bmatrix}$
B A Gnd	NO COM NO COM
RS 485	Relay 1 Relay 2

Model	Key features
END25RS	RS485 interface + 2 Pulses outputs, accuracy class 0.5s
END25ETH	Ethernet + data logging, accuracy class 0.2s

Full version of user's manual available here







ETI Elektroelement d.o.o. 1411 Izlake Slovenia

АНАЛІЗАТОР ПАРАМЕТРІВ МЕРЕЖІ END25

ІНСТРУКЦІЯ ПО ВИКОРИСТАННЮ ШВИДКИЙ СТАРТ Повна версія інструкції доступна

на сайті www.eti.ud

1. ВСТУП

END25 - це багатофункціональний цифровий панельний аналізатор розмірами 96 х 96 мм, що монтується на панелі або дверцятах шафи і який вимірює важливі електричні параметри в 3-фазній 4-провідній / 3-фазній 3-провідній / 1-фазній мережах та замінює собою кілька аналогових панельних лічильників. Прилад вимірює електричні параметри, такі як напруга змінного струму, сила струм, частота, потужність, енергія (активна / реактивна / повна), кут зсуву фаз, коефіцієнт потужності, окремі гармоніки і багато іншого. У приладі використовується технологія точних вимірювань (всі вимірювання напруги і струму відповідають дійсним середньоквадратичним значенням до 31-ї гармоніки) з РК-дисплеєм який підсвічується.

ALLAN)

ŜDD

*0000 19672 ×m

Аналізатор можна налаштувати і запрограмувати на об'єкті змінюючи наступні параметри: значення напруг для первинної і вторинної обмоток ТН, значення струмів для первинної і вторинної обмоток TC, 3-фазна 4-провідна, 3-фазна 3-провідна, 1-фазна 2-провідна системи.

На передній панелі є три кнопки за допомогою яких користувач може переглядати різні екрани і налаштовувати прилад. На передній панелі також є імпульсний червоний світлодіод, який блимає зі швидкістю, яка пропорційна вимірюваній потужності.



2. ЕКРАНИ ВИМІРЮВАННЯ І ЕКРАНИ ЕНЕРГІЇ / ЛІЧИЛЬНИКІВ

У нормальному режимі роботи користувачу одночасно доступні два екрани:

1. Один із Екранів вимірювань - екран з 1 по 36 Таблиці 1.

Ці екрани можна прокручувати по одному в порядку зростання, натискаючи кнопку «Вгору», і в порядку спадання, натискаючи «Вниз». Деякі важливі екрани детально розкриті в розділах 2.1 - 2.6 **.

2. Один з Екранів обліку енергії / лічильників - екран 37-62 Таблиці 1. Ці екрани можна прокручувати по одному в порядку зростання, натиснувши тільки кнопку «Enter», щоб перегорнути знову в тому ж порядку. Деякі важливі екрани розкриті в розділах 2.7-2.9 **

Графіки завантаження

Графіки завантаження показують вхідний струм у відсотках від значення струму первинної обмотки трансформатора струму. Ця індикація доступна для всіх екранів вимірювань. Наприклад, припустимо, що струм первинної обмотки TC встановлений на 5 А, тоді вхідний струм 2,5 А означає 50%, як показано нижче.

** - див. повну версію інструкції користувача.



Вілсутність піній. 1110 вказують ня у відсотках, означає, що вхідний струм не перевищує 20% від струму ної обмотки ТС перв

Індикатор чергування фаз

Він вказує обертання вхідних векторів фаз: за чи проти годинникової стрілки. Якщо напруга відсутня або послідовність фазне відповідає І 123 або І 321 індикація юслідовності фаз не відображається

 Послідовність проти С ргодинникової стрілки для L321) годинниковою стрілкою

ТАБЛИЦЯ 1: Екрани вимірювань і енергії / лічильників для базової версії END25

для L123

N≌	R	1	На екрані			Πo Modbus		
комірки	Параметри		3Ф 3П	1Ф 2П	3Ф 4П	3Ф 3П	1Φ 2I	
1	Напруга мережі / струм / потужність / частота	✓	✓	✓	 ✓ 	~	✓	
2	Фазна напруга	✓	×	×	✓	×	×	
3	Лінійна напруга	✓	✓	×	✓	√	×	
4	Струм, струм в нейтралі	 ✓ 	√*	×	✓	√*	×	
5	ТНD фазної напруги	×	×	×	×	×	×	
6	ТНD фазного струму	×	×	×	×	×	×	
7	Фаза L1 - повна / активна / реактивна потужність	✓	×	×	 ✓ 	×	×	
8	Фаза L2 - повна / активна / реактивна потужність	✓	×	×	 ✓ 	×	×	
9	Фаза L3 - повна / активна / реактивна потужність	✓	×	×	✓	×	×	
10	Кут зсуву фаз	×	×	×	×	×	×	
11	Повна потужність / струм, середнє значення	×	×	×	×	×	×	
12	Ємнісна, індуктивна потужність, середнє значення	×	×	×	×	×	×	
13	Споживана активна потужність, середнє значення	×	×	×	×	×	×	
14	Генерована активна потужність, середнє значення	×	×	×	×	×	×	
15	Мах повна потужність / max струм, середнє значення	×	×	×	×	×	×	
16	Мах ємнісна, індуктивна потужність, середнє значення	×	×	×	×	×	×	
17	Мах споживана активна потужність, середнє значення	×	×	×	×	×	×	
18	Мах генерована активна потужність, середнє значення	×	×	×	×	×	×	

ТАБЛИЦЯ 1: Екрани вимірювань і енергії / лічильників для базової версії END25

N≌	Параметри		На екрані			Πo Modbus		
омірки			3Ф 3П	1Φ 2Π	3Ф 4П	3Ф 3П	1Φ 2	
10	Мах повна потужність / тах струм, середнє значення,	~	~	~	~	~	~	
15	додатковий лічильник	<u>^</u>	^	^	^	^	^	
20	Мах ємнісна, індуктивна потужн., середнє значення,	~	~	~	~	~	~	
20	додатковий лічильник		-	<u>^</u>	~	~	~	
21	Мах споживана активна потужність, середнє значення,	×	×	×	×	×	×	
21	додатковий лічильник							
22	Мах генерована активна потужність, середнє значення,	×	×	×	×	×	×	
	додатковий лічильник							
23	3-фазна синхронна швидкість генератора / частота	✓	✓	✓	✓	✓	✓	
24	3-фазна повна / активна / реактивна потужність	✓	✓	✓	√ (✓ ×	×	
25	З-фазний кут зсуву фаз, коефіцієнт потужності		✓	✓	✓	✓	×	
26	Міп 3-фазна напруга / міп 3-фазний струм	×	×	✓	✓	✓	×	
27	мах 3-фазна напруга / тах 3-фазнии струм	×	×	✓	×	✓	×	
28	3-фазне ТНО напруги / струму	×	~	v v	~	~	~	
29	Зміна напрямку струму	_ √	×	✓	×	×	×	
30	Помилка чергування фаз	✓	~	×	~	~	×	
31	індикатор оориву фази	×	×	×	×	×	×	
32	і одинник реального часу	~	×	~	~	~	~	
33	індивідуальні гармоніки напруг	×	×	×	×	×	×	
34	індивідуальні гармоніки струмів	×	×	×	×	×	×	
35	Таимер 1 кількості циклів, затримка включення, виключення	×	×	×	×	✓	×	
30	Таимер 2 кількості циклів, затримка включення, виключення	~	V	~	✓	✓	×	
3/	Споживана активна енертя (надлишок)	×	×	×	✓	✓	×	
30	Споживана активна енергія	~	×	~	✓	✓	×	
39	I енерована активна енергія (надлишок)	×	×	×	×	✓	×	
40	I енерована активна енергія	~	×	~	×	✓	×	
41	Смнісна реактивна енергія (надлишок)	×	×	×	×	✓	×	
42	Смнісна реактивна енергія		v	~	•	•	*	
43	пндуктивна реактивна енертія (надлишок)	*	*	×	¥	•	¥	
44	пндуктивна реактивна енертія	v	v	~	v (v (×	
45	Повна енергія (надлишок)	×	×	×	✓	✓	×	
40	Повна енергія	× (× (× (✓	✓	×	
47	Личильник часу росоти навантаження	¥	× (¥	*	¥	*	
40	Личильник часу росоти	¥	¥	¥	¥	¥	*	
49 50	Пичильник зникнення напруги живлення приладу	*	*	*	*	*	*	
51	Споживана активна енергія (надлишок), додат. лічильник	÷	÷	÷	Ŷ	×	÷	
52	Споживана активна енергія, додат. лічильник	~ ~	~	~	×	~ ×	~ *	
53	Генерована активна енергія (надлишок), додат. лічильник	~	~	~	~	~	~	
54	Смијсиа реактивна енергія (наллишок) полат пічильник	÷	÷	÷	×	×	÷	
55	Сминска реактивна сперти (падлишек), додат. личиленик	~ ~	~	~	×	~ *	~ *	
56	Смпюла реактивна енергія, додаї, личильник	~ ~	~	~	~ *	~ *	~ *	
57	падуктивна реактивна енергія (падлишок), додат. лічильник	- ~	~	~	~ *	~ *	~	
58	Пориз енергія (изламшок) полат пічильник	×	×	×	×	×	- ×	
59	Повна енергія подат пічильник	×	×	×	×	×	×	
60	Пічильник часу роботи навантаження, додат, пічильник	×	×	×	×	×	×	
61	Лічильник часу роботи, додат, пічильник	×	×	×	×	×	×	
	Лічильник зникнення напруги живлення прилалу							
62	лолатковий пічипьник	×	×	×	×	×	×	

Параметри вихідних контактів Æ 11 11 <u>0 10</u> IFE, EFE, Col ConF SEL (Baőip kostójirypauj (En, d-En) PuLS dur (1 0↓ ↓0 on dELY 00 tiME (Час увімкнення) 00:00 to 23:59 1 1 Û↓ †Û OFF dELY (3arpsinska Ha BRINK 0001 to 9999 (7000 10 9999) імпульсів) 10 100 100 off tiME 11 II (Роц 3.2.5.1.1.3)* **1** 1 1 1 guit (Вихід з меню 01 ↓ ↓ J on dELY атримка на увімк. 1с до 9999с ÎI↓ †J ©<u>ut</u>in Î : Кнопка ВГОРУ : Кнопка ENTER Л : Кнопка ВНИЗ ዾዀ Параметри реєстрації даних 📿 Û. Û (Batuin 3 MCHBO 28 M (Poun 3.2.6.4) див. повну версію інструкції користувача.

(1) Для екранів користувача, які можна обирати, доступні тільки екрани за номерами від 1 до 32. (2) Енергія на дисплеї автоматично змінюється. Подробиці див. в розділі 4.1 **

*В системі 3Ф3П струм в нейтральному провіднику не відображається, відображаються тільки лінійні струмі **Див. повну версію інструкції користувача.



3. ПРОГРАМУВАННЯ

Наступні розділи містять покрокові інструкції налаштування аналізатора параметрів мережі відповідно до індивідуальних вимог користувача. Для доступу до екранів налаштування одночасно натисніть і утримуйте кнопки ▲ «ВГОРУ» та ▼ «ВНИЗ» протягом 5 секунд. Це переведе користувача на етап введення паролю захисту (Розділ 3.1)**

3.1. Захист паролем

Можна активувати захист паролем для запобігання несанкціонованого доступу до екранів налаштування, якщо захист паролем за замовчуванням не активовано. Захист паролем відбувається шляхом вибору чотиризначного числа, відмінного від 0000. Встановлення пароля 0000 відключає захист паролем.

	EodE	Введення паролю. Вибір першої цифри. Натисніть кнопку «▲» для прокручування значення першої цифри від 0 до 9, значення повертається з 9 на 0, або кнопку	
		«▼» - для прокручування значення першої цифри від 9 до 0, значення повертається з 0 на 9.	
		Натисніть кнопку «🛥», щоб перейти до наступної цифри.	3.2
	EodE	В особливому випадку, коли пароль «0000», натискання кнопки « , при запиті першої цифри, призводить до переходу на екран підтвердження пароля, а повторне натискання кнопки	
	0000 donE	«■» робить екрани налаштування доступними для користувача. Але замість натискання «■», якщо натиснути кнопку «▲» або ««», користувач потрапляє на етап введення «Новий / змінити пароль».	
	EodE	Введення паролю. Перша цифра введена, вибір другої цифри. Натисніть кнопку «▲» для прокручування значення другої цифри від 0 до 9, значення повертається з 9 на 0, або кнопку ««» - пля проклаивания значения пригої цифри від 0 до 0. зачанни	3.2
	[повертається з 0 на 9. Натисніть кнопку «ш», щоб перейти до наступної цифри.	
J		′] Введення паролю. Друга цифра введена, вибір третьої цифри.	
	EodE	Натисніть кнопку «▲» для прокручування значення третьої цифри від 0 до 9, значення повертається з 9 на 0, або кнопку	
	(3	«•» - для прокручування значення третьої цифри від 9 до 0, значення повертається з 0 на 9.	3.2
		Натисніть кнопку «💼», щоб перейти до наступної цифри.	
	EodE	Введення паролю. Третя цифра введена, вибір четвертої цифри. Натисніть кнопку «▲» для прокручування значення четвертої цифри від 0 до 9, значення повертається з 9 на 0, або кнопку	
	134-	«▼» - для прокручування значення четвертої цифри від 9 до 0, значення повертається з 0 на 9.	
		Натисніть кнопку « — », щоб перейти до перевірки пароля.	3.2
	٢٠٩٢	Введення паролю. Четверта цифра введена, очікування	
	LOOC	під вердження пароля.	
	(342		
1		1	



3.2.5. Вибір параметра вихідних контактів



Цей екран дозволить користувачеві встановити параметри

релейних вихідних контактів Натискання кнопки «) дозволяє користувачеві вибрати налаштувати параметри вихідних контактів (Розділ 3.2.5.1)**. Натискання кнопки «▲» призведе до переходу на екран «Вибір параметра журналу даних» (Розділ 3.2.6)**, а натискання кнопки «• - до екрану «Вибір параметра скидання» (Розділ 3.2.4)**.

3.2.6. Вибір параметра журналу даних

SEL drer Lou SEŁuP

5EL

SEL.P

d ,SP PRrR

завантаження» Натискання кнопки « ж дозволяє користувачеві вибрати налаштувати параметри журналу даних (Розділ 3.2.6.1)**.

Натискання кнопки «▲» призведе до переходу на екран «Параметри екрану» (Розділ 3.2.7)**, а натискання «▼» - до екрану «Вибір параметра вихідних контактів» (Розділ 3.2.5)**.

3.2.7. Параметри екрану

Цей екран дозволить користувачеві отримати доступ до різних функцій, таких як «Підсвічування», «Екрани користувача», «Контрастність» і «Додаткові параметри».

Натискання кнопки «) дозволяє користувачеві вибрати налаштувати параметри екрану (Розділ 3.2.7.1)**. Натискання кнопки «▲» призведе до переходу

«Налаштування годинника реального часу» (Розділ 3.2.8)**. а натискання кнопки «▼» - до екрану «Вибір параметра журналу даних» (Розділ 3.2.6)**

3.2.8. Налаштування годинника реального часу



Цей екран дозволить користувачеві отримати доступ до таких функцій, як «Встановити дату» та «Встановити час» Натискання кнопки « » дозволяє користувачеві вибрати дату

час (Розділ 3.2.8.1)* Натискання кнопки «▲» призведе до переходу на екран «Скидання до заводських налаштувань» (Розділ 3.2.9)**, а натискання кнопки «•» - до екрану «Параметри екрану» (Розділ 3.2.7)**

3.2.9. Скидання до заводських налаштувань



Цей екран дозволяє користувачеві скинути аналізатор до заводських налаштувань (Розділ 3.2.9.1) **. Натискання кнопки «▲» призведе до переходу на екран «Вихід з меню» (Розділ 3.2.10)**, а натискання кнопки «▼» - до екрану «Налаштування годинника реального часу» (Розділ 3.2.8)**. Для забезпечення безг

3.2.10. Вихід з меню



Цей екран дозволить користувачеві вийти з меню налаштувань (Розділ 3.2.10.1) ** Натискання кнопки « жозволяє користувачеві вийти з меню

налаштувань і повернутись до екрану вимірювань. Натискання кнопки «▲» призведе до переходу на екран «Вибір системного параметра» (Розділ 3.2.1)**, а натискання кнопки «•» - до екрану «Скидання до заводських налаштувань» (Розділ 3.2.9)**.

1-3 мм для затискачів

1-6 мм для гвинтів

** - див. повну версію інструкції користувача.

4. МОНТАЖ

Прилад монтується за допомогою утримуючих затискачів. Вставте аналізатор в монтажний отвір розміром 92 x 92 мм та зафіксуйте його за допомогою чотирьох затискачів, що розміщені з двох боків приладу. При необхідності можна використовувати два додаткові гвинтові тримачі, як показано на рисунку.

Передня частина корпусу має ступінь захисту ІР54. Додатковий захист панелі можна отримати. частина корпусу має ступінь захисту використовуючи додаткову прокладку для отвору на панелі. Клеми на задній частині приладу повинн бути захищені від контакту з рідинами.

Аналізатор повинен працювати за відносно стабільної температури навколишнього середовиша в діапазоні від -10° до 55°С. Вібрації повинні бути зведені до мінімуму і прилад не слід встановлювати в місцях з прямими сонячними променям

VBaral

1. В інтересах безпеки і функціональності цей прилад повинен бути встановлений кваліфікованим персоналом з дотриманням всіх місцевих норм техніки безпеки. 2. На клемах даного приладу присутня небезпечна для життя напруга. Перед спробою підключення або відключення переконайтеся, що всі джерела живлення знеструмлені 3. Прилад не має внутрішніх запобіжників. Для захисту приладу від пошкоджень викликаних надмірним струмом, слід використовувати зовнішні запобіжники.

4.1. Вимоги ЕМС

Цей прилад був розроблений відповідно з сертифікацією директив ЄС щодо електромагнітної сумісності для приладів, що працюють у промисловому середовищі: 1. Екрановані вихідні і вхідні кабелі з низьким рівнем сигналу або передбачені засоби для встановлення компонентів зменшення радіочастот, таких як феритові сердечники мережеві фільтри і т. д., у випадку виникнення проблем з радіочастотними полями. Примітка: рекомендується встановлювати чутливі електронні прилади, які виконують критично важливі функції, в корпусах з електромагнітною сумісністю, які мають захист від електромагнітних перешкод, що можуть викликати порушення роботи.

2. Уникайте прокладання вимірювальних кабелів поруч з кабелями і виробами, які є 6. ДОДАТКОВИЙ МОДУЛЬ або можуть бути джерелами перешкод.

 Щоби захистити прилад від незворотного пошкодження, імпульсні пікові перенапруги повинні бути обмежені до 2 кВ. Гарною практикою ЕМС є зменшення стрибків напруги до 2 кВ в джерелі. Прилад був розроблений з автоматичним відновлення в разі високого рівня перехідних процесів. Однак у разу надзвичайно високих перенапруг може знадобитися тимчасове відключення джерела живлення на період більше 5 секунд для відновлення належного функціонування. Трансформатори струму, що використовуються з аналізатором, повинні мати

заземлення вторинної обмотки. 4. При використанні цього приладу необхідно завжди дотримуватися запобіжних заходів захисту від електростатичного розряду.



4.3. Підключення провідників

З'єднання вхідних контактів виконуються безпосередньо на гвинтових клемах з непрямим тиском на провід. Нумерація чітко позначена на клемі. Вибір кабелю повинен відповідати місцевим нормам. Для вхідних клем струму і напруги дозволено використовувати одножильний кабель перерізом до 4 мм² (12AWG) або багатожильний кабель перерізом до 2,5 мм².

Примітка: рекомендується використовувати кабель з наконечником

4.4. Живлення приладу

Прилад повинен живитися від окремого джерела енергії, однак можна забезпечити живлення від вимірювальної мережі, за умови, що напруга перебуває в діапазоні, допустимому для застосування в якості джерела живлення

4.5. Запобіжники

Рекомендується, щоб усі лінії напруги були захищені запобіжниками з високою вимикальною здатністю з номіналом в 1 ампер.

Для забезпечення безпеки вторинні обмотки трансформаторів струму повинні бути заземлені відповідно до вимог монтажу.

5. СХЕМИ ПІДКЛЮЧЕННЯ





7. ТЕХНІЧНІ ХАРАКТЕРИСТИКИ

Іережа	3-фазна 3- або 4-провідна, 1-фазна програмується безпосередньо на об'єкті
имірювання	
Іомінальна вхідна напруга АС, середньоквадратична)	Лінійна напруга 100V - 600 V L-L Фазна напруга 57,7V - 346,4 V L-N програмується безпосередньо на об'єкті
апруга первинної обмотки ТН	Від 100V L-L до 1200 kV L-L програмується безпосередньо на об'єкті
lапруга вторинної обмотки TH	Від 100V L-L до 600 V L-L програмується безпосередньо на об'єкті
Іаксимальна тривала вхідна напруга Іомінальне вхідне навантаження по апрузі	120% від номінального значення <0,3 ВА на фазу (при номінальній U - 240 В)
юмі́нальний вхідний струм	
аксимальний тривалий вхідний струм Іомінальне вхідне навантаження по труму	<0,3 А на фазу
струм первинної обмотки ТС	Від 1А до 9999А програмується безпосередньо на об'єкті
трум вторинної обмотки ТС	1А/5А програмується безпосередньо на об'єкті
ндикація перевантаження	"-OL-" >121% від номінального значення (для напруги та струму)
итримуване перевантаження	
хід по напрузі	2 х номінального значення (1-секундний повтор 10 разів з 10-секундним інтервалом)
хід по струму	2 х номінального значення (1-секундний повтор 5 разів з 5-секундним інтервалом)
овнішнє джерело живлення	
апазон напруги джерела живлення	100V - 550V AC/DC
юмінальна напруга джерела живлення	230V AC/DC 50 HZ
астота джерела живлення авантаження по напрузі з додатковим юдулем	<6 VA (за номіналом)
lавантаження по напрузі з модулем ithernet	<8 VA (за номіналом)
обочі діапазони вимірювань	00 400 % -i
апруга	20 120 % від номінального значення
ірум Іусковий струм	
actota	45 - 66 Hz
оефіцієнт потужності	0,5 індуктивний 1 0,8 ємнісний
агальна гармонійне спотворення	50% до 15-ї гармоніки 10% до 31-ї гармоніки
очність вимірювання забезпечується емпература (згідно IEC 62053-21)	при 23°C ± 2°C
	±0.5 % від номінального значення
бтрум	±0,5 % від номінального значення
астота	±0,2 % від середнього значення
ктивна потужність	±0,5 % від номінального значення
еактивна потужність	±1,0 % від номінального значення +0.5 % від номінального значення
оефіцієнт потужності / Кут зсуву фаз	±3°
ктивна енергія	згідно IEC 62053-22 (0,5 с)
	згідно IEC 62053-22 (0,2 с) (опціонально)
еактивна енергія	Клас 2 згідно ІЕС 62053-23
Ювна енертя НD (Напруга / Струм)	+3.0 %
очність вимірювання (для опціональн Іапруга	их 0,2 с) +0.2 % від номінального значення
Струм	±0,2 % від номінального значення
астота	±0,2 % від середнього значення
ктивна потужність	±0,2 % від номінального значення
овна потужність кран	±0,2 % від номінального значення
К-дисплей з підсвічуванням	
ас відгуку на введення	Приблизно 1 секунда
правлиня нтерфейс користувача	3 кнопки
Відповідність стандартам	
MC	IEC 61326 - 1 : 2012
тійкість	IEC 61000-4-3. 10V/m min – Рівень 3 Промисловий низький рівень
езпека	IEC 61010-1-2010, використовується постійне підключення
лупнь захисту ГР	IEC 00029



Slovenia

Ступінь забруднення	2
Категорія перенапруги	111
Ізоляція	
Клас захисту	2
Випробування високою напругою	
Вхідні клеми+клеми живлення до поверхні	4 kV середнє значення, 50 Hz, 1 хв
Вхідні клеми до інших клем	3,3 kV середнє значення, 50 Hz, 1 хв
Параметри навколишнього середовиш	a
Робочий діапазон температур	Від -10 до +60°С
Температура зберігання	Від -20 до +70°С
Відносна вологість	0 90% (без утворення конденсату)
Час нагріву приладу	Мінімум 3 хвилини
Ударний імпульс (згідно IEC 60068-2-27)	Напівхвиля синусоїди, пікове прискорення 30 gn (300 м/с ²), тривалість - 18 мс
Вібрація	10 150 10 Hz, амплітуда 0,15 мм
Кількість циклів розгортки	10 на вісь
Параметри корпусу	
Передня частина	IP54
Задня частина (клеми)	IP20
Розміри	
Розмір передньої частини	96 мм x 96 мм DIN 43718
Виріз в панелі	92 мм+0,8 х 92 мм+0,8
Загальна глибина	75 мм
Товщина панелі	1 - 3 мм для затискачів 1 - 6 мм для гвинтів
Maca	320 грам
Параметри імпульсних вихідних конта	ктів
Тип контакту	1 NO
Потужність комутації	5 A, 240 V AC
Дільник частоти імпульсів за замовчуванням	1 на Втгод (до 3600 Вт)
	1 на кВтгод (до 3600кВт)
	1 на МВтгод (від 3600 кВт до 30000 кВт)
Дільник частоти імпульсів	Програмується безпосередньо на об'єкті
10	1 на 10 Втгод (до 3600 Вт)
	1 на 10 кВтгод (до 3600кВт)
	1 на 10 МВтгод (від 3600 кВт до 30000 кВт)
100	1 на 100 Втгод (до 3600 Вт)
	1 на 100 кВтгод (до 3600кВт)
	1 на 100 МВтгод (від 3600 кВт до 30000 кВт)
1000	1 на 1000 Втгод (до 3600 Вт)
	1 на 1000 кВтгод (до 3600кВт)
	1 на 1000 МВтгод (від 3600 кВт до 30000 кВт)
Тривалість імпульсу	60 мс, 100 мс або 200 мс

Примітка:

Парам Протон

Швидк

Парніс

Ethern

Імпул

Більш докладну інформацію див. в таблиці 4 **. 2. Вищевказані умови також відносяться для реактивної і повної енергії.

етри ModBus (RS 485)	
юл	ModBus (RS485)
сть передачі	4.8k, 9.6k, 19.2k, 38.4k, 57.6kbps
ТЬ	Непарний або парний з 1 стоповим бітом або "None" з 1 або 2 стоповими бітами
et	Доступ до Ethernet по протоколу Modbus TCP/IP
сний вихід	

Імпульсний світлодіод

Для тестування

8. ПІДКЛЮЧЕННЯ ДОДАТКОВОГО ІМПУЛЬСНОГО ВИХОДУ / RS485 (вид аналізатора ззаду)

Вихідні контакти та контакти RS485

• • •	0 0 0 0
B A Gnd RS 485	NO COM NO COM Вихід 1 Вихід 2

Модель	Особливість
END25RS	Інтерфейс RS485 + 2 імпульсних групи контактів, клас точності 0,5s
END25ETH	Ethernet + реєстрація даних, клас точності 0,2s

Повна версія і інструкція користувача різними мовами доступні на сайті www.etigroup.eu і по QR-коду





ETI Elektroelement d.o.o. 1411 Izlake

Tel. : +386 03 56 57 570 e-mail: eti@eti.si



UPORABNIŠKI PRIROČNIK

Celotna navodila so dostopna tu *www.etigroup.eu*

1. UVOD

Večnamenski instrument dim. 96 x 96 mm se vgradi v nadzorno ploščo in meri vnesene pomembne električne parametre v 3 ph 4-žičnem/3-žičnem/1ph omrežju in nadomešča več analognih števcev. Meri električne parametre, kot so izmenična napetost, tok, frekvenca, moč energija (delovno/induktivno/navidezno) fazni kot faktor moči posamezne harmonike in še velikja (delovinduktivionavležno), jažin od. jako noto, posanezne namonke in še veliko več. Instrument vključuje natančno merilno tehnologijo (meri vse napetosti in tokovr preko "True RMS" do 31. harmonike) z osvetljenim LCD zaslonom.

ALLAN)

'0000 (9672 ×m

Inštrument je lahko konfiguriran&programiran za sledeče velikosti: PT Primarn, PT Sekundarna, CT Primarna, CT Sekundarna 3 fazna 3W, 3 Fazna 4W, 1 fazna 2W sistem.

Inštrument ima tri (3) tipke, s katerimi se uporabnikih pomika po različnih menijih in kon-figurira izdelek. Inštrument ima rdečo signalno "ledico", ki utripa sorazmerno z izmerjeno močjo.



2. ŠTEVEC MERITEV IN ENERGIJ

V normalnem delovanju, sta uporabniku predstavljena dva prikaza istočasno.

1. En od MERILNIH MENIJEV na zaslonu je prikazan v TABELI 1.

Po prikazih se lahko pomika po naraščajočem vrstnem redu, tako da se pritisne tipko "gor", v padajočem vrstnem redu tako, da pritišne tipko "dol".

Nekaj pomembnih prikazov je razloženih v poglavju 2.1 in 2.6. *

2. Eden od ENERGIJSKI/ŠTEVEC PRIKAZOV od prikazov 37 - 62 v Tabeli 1. Ti prikazi se prikazujejo posamično po vrstnem redu ob pritiskanju tipke "POTRDI". Nekaj pomembnih prikazov je razloženih/prikazanih v poglavju 2.7 - 2.9 **.

NALOŽI GRAFIČNI PRIKAZ

Naloži grafiko označuje vhodni tok kot odstotek vrednosti CT primarna vrednost. Ta indikator ie možen za vse meritve zaslona

Primer, če je CT Primarni nastavljen na 5A, tok na vhodu pa je 2,5A, bo na zaslonu prikazana vrednost 50%. Risba spodaj.

** - Glej celotno različico navodil.



Indikator zaporedja faz

Prikazuje smer vrtilnega napetostnega polia:

Če ni indikacijskih črtic je vhodni tok manjši od 20% CT Primarne vrednosti

V smeri/proti smeri urnih kazalcev. V primeru da vhodne faze niso ne L123 ne L321 indikator ne prikazuje nič.

V smeri urnih kazalcev

Proti smeri urnih kazalcev **†**L321.

TABELA 1: Osnovni prikazi analizatorja mreže END25

L123

Screen	Barrantan	On Display			On Modbus		
No.	No. Parameters		3P 3W	1P 2W	3P 4W	3P 3W	1P 2W
1	System Voltage/ Current/ Power/ Frequency	✓	 ✓ 	√	√	√	 ✓
2	L-N Voltage	√	×	×	~	×	×
3	L-L Voltage	✓	 ✓ 	×	~	~	×
4	Current, Neutral Current	√	√*	×	√	√*	×
5	Phase Voltage THD	×	×	×	×	×	×
6	Phase Current THD	×	×	×	×	×	×
7	L1 phase VA/VAr/W	√	×	×	√	×	×
8	L2 phase VA/VAr/W	✓	×	×	 ✓ 	×	×
9	L3 phase VA/VAr/W	✓	×	×	~	×	×
10	Phase Angle	×	×	×	×	×	x
11	VA/A demand	×	×	×	×	×	×
12	Capacitive, Inductive VAr demand	×	×	×	×	×	×
13	W IMP demand	×	×	×	×	×	×
14	W EXP demand	×	×	×	×	×	×
15	Max VA/A demand	×	×	×	×	×	×
16	Max Capacitive, Inductive VAr demand	×	×	×	×	×	×
17	Max W IMP demand	×	×	×	×	×	×
18	Max W EXP demand	×	×	×	×	×	×

Screen	reen		On Display			On Modbus		
No. Parameters		3P 4W	3P 3W	1P 2W	3P 4W	3P 3W	1P 2W	
19	Old Max VA/A demand	×	×	×	×	×	×	
20	Old Max Capacitive, Inductive VAr demand	×	×	×	×	×	×	
21	Old Max W IMP demand	x	×	x	x	x	x	
22	Old Max W EXP demand	×	×	×	×	×	×	
23	Sys RPM/Frequency	✓	~	✓	✓	~	~	
24	Sys Active/Reactive/Apparent power	✓	✓	~	~	~	~	
25	Sys Apparent, Reactive (Power), Phase angle, PF	✓	✓	√	√	✓	~	
26	Min Sys Voltage / Current	√	✓	√	√	√	√	
27	Max Sys Voltage / Current	✓	✓	~	~	~	~	
28	Sys %THD Voltage /Current	✓	✓	√	√	✓	~	
29	Current Reversal	√	×	√	×	×	×	
30	Phase rotation error	✓	✓	×	~	~	×	
31	Phase absent indication	√	×	×	×	×	×	
32	RTC	✓	✓	✓	√	✓	✓	
33	Individual harmonics V	×	×	×	×	×	×	
34	Individual harmonics A	×	×	×	×	×	×	
35	Timer1 No. of Cycles, ON, OFF delay	✓	✓	✓	✓	✓	✓	
36	Timer2 No. of Cycles, ON, OFF delay	✓	✓	√	√	~	~	
37	Active Energy Import(Overflow)	×	×	×	√	√	√	
38	Active Energy Import	√	✓	√	√	✓	√	
39	Active Energy Export(Overflow)	×	×	×	✓	✓	✓	
40	Active Energy Export	√	✓	√	√	√	√	
41	Reactive Capacitive energy (Overflow)	×	×	×	√	√	√	
42	Reactive Capacitive energy	✓	✓	√	√	√	√	
43	Reactive Inductive energy (Overflow)	×	×	×	√	√	√	
44	Reactive Inductive energy	√	√	√	√	√	√	
45	Apparent energy(Overflow)	×	×	×	✓	✓	✓	
46	Apparent energy	√	√	√	√	√	√	
47	Run hour	✓	✓	✓	✓	✓	✓	
48	On hour	✓	✓	✓	✓	✓	✓	
49	No. of interrupts	√	✓	√	√	✓	✓	
50	Old Active energy Import(Overflow)ĸ	×	×	×	×	×	×	
51	Old Active energy Import	×	×	×	×	×	×	
52	Old Active energy Export(Overflow)	×	×	×	×	×	×	
53	Old Active energy Export	×	×	×	×	×	×	
54	Old Reactive Capacitive energy (Overflow)	×	×	×	×	x	x	
55	Old Reactive Capacitive energy	×	×	×	×	×	×	
56	Old Reactive Inductive energy (Overflow)	×	×	×	×	×	×	
57	Old Reactive Inductive energy	×	×	×	×	×	×	
58	Old Apparent energy(Overflow)	×	×	×	×	×	×	
59	Old Apparent energy	×	×	×	×	×	×	
60	Old Run hour	×	×	×	×	×	×	
61	Old On hour	×	×	×	×	×	×	
62	Old no. of interrupts	×	×	×	x	x	x	

TABELA 1: Osnovni prikazi analizatorja mreže END25

(1) Prikaz samo (s številkami) 1 -32 so možne samo za določene Prikaze (2) Energije na prikazu je samodejnem razponu za podrobnosti glej Poglavje 4.1.*

* Opozorilo: V 3P3Wsistemu, so prikazani le fazni tokovi, tok nevtralnega vodnika ni prikazan * - Glej razširjena navodila/priročnik





3. PROGRAMIRANJE

EodE 1342

Poglavje vas vodi korak za korakom skozi proceduro konfiguracije Analizatorja mreže, glede na uporabnikove zahteve. Za dostop do nastavitev zaslona, pritisnite in držite tipki ▲ «GOR» in ▼ «DOL» istočasno za cca 5s. To bo uporabnika vodilo v vstopno stopnjo zaščite z geslom (Poglavie 3.1) **.

Zaščita z geslom onemogoča dostop nepooblaščenim dostop do nastavitev. Geslo je lahko katerekoli štirimestno število (razen 0000). Nastavitev gesla s št. 0000 izklopi geslo.

Vnesite geslo, vnesena četrta številka, čakate na preverjanje/ potrditev gesla.

Defeditors and a	
	S pritiskom tipke «▲» ali «▼» se premakneš v Meni «NOVO/MENJAVA GESLA»
	S pritiskom tipke « ■ » napreduješ v Meni izbiral zaslona (Glej poglavje
ane Brab	3.2.)**.
Napačno geslo	
EodE	The unit has not accepted the Password entered.
	S pritiskom tipk «▲» ali «▼» se prestavimo v fazo potrditev gesla.
Err	Pritiskom na (IIII) tipko zapustite meni Geslo in se vrnete v način odčitavanja meritev.
Novo / menjava gesla	1
EodE	Vnos prvega znaka.
0000	S pritiskom gumba «▲» in «♥» za nastavitev vrednosti od 0-9 in od 9-0.
Ēdīk	S pritiskom gumba «••» napredujemo na prvi znak in ga nastavimo na vrednost 2.
	Novo/menjava gesla vnos znaka.
	S pritiskom gumba «▲» in «▼» za nastavitev vrednosti od 0-9 in od 9-0.
2000 Ed 12	S pritiskom gumba «■» napredujemo na naslednji znak in ga nastavimo na vrednost 1
EodE	Novo/menjava gesla vnos znaka.
2 (00	S pritiskom gumba «▲» in «♥» za nastavitev vrednosti od 0-9 in od 9-0.
Edik	na vrednost 5.
	Novo/menjava gesla vnos znaka.
	S pritiskom gumba «▲» in «▼» za nastavitev vrednosti od 0-9 in od 9-0.
	S pritiskom gumba «🗩» snapredujemo na naslednji znak in ga nastavimo na vrednost 3.
Potrditev novega ges	la
EodE	S pritiskom gumba «▲» in «♥» se prestavi v «NOVO/MENJAVA GESLA».
2153	3.2.)**.
325	
3.2. Uzbira men	ija
3.2.1. Meni izbir	e parametrov
5EL 545	«CT Ratio», «PT Ratio» S pritiskom tipke «■» omogoča uporabniku da spremeni sistem parametrov
P Ŕŕ R	(Glej poglavje 3.2.1.1 do 3.2.1.12)**. S pritiskom na tipko «a» se premaknete v meni «Komunikacijski pormetar (Cleita poglavie 2.2.2) in pritickom tipko tipko «mu promokrata
SEŁup	v meni «KONEC NASTAVITEV» (Glej poglavje 3.2.9).
3.2.2. Komunika	ncijski parameter Meni se uporablja za izbiro različnih komunikaciiskih parametrov, kot so
<u>5</u> EL	«Izbira naslova», «Izbira paritete RS485», «Hitrost prenosa RS485» itd. S pritiskom na tipko «==» lahko uporabnik nastavi drugačne komunikacijske
	parametre (ule) poglavje 3.2.2.1 do 3.2.2.3)**. S pritiskom tipke «▲» se premaknete v meni «Ponastavitev parametrov» (ob komunikaciji preko ethernetnena modula, se nadaljuje na izbiro IP poslova
SEŁup	glejte poglavje 3.2.3) (Glejte poglavje 3.2.4) in pritiskanje tipke ∢▼» napre- dujete v meni «Izbira sistemskih parametrov» (Glejte poglavje 3.2.1).
3 2 3 Izhira par	ametra IP
	Meni se uporablja za izbiro različnih komunikacijskih parametrov, kot so
SEL IP	«IF auresa», «IZUITA maske podomrezja», «privzeti prehod izbira», izbira vrat strežnika itd. S pritiskom na tipko «=» lahko uporabnik nastavi različne Ethernet
PRrR	parametre (Glejte poglavje 3.2.3.1 do 3.2.3.5)**. S pritiskom na tipko «▲» se premaknete v poglavje «Ponastavi
	parametre»(Glejte poglavje 3.2.4) in s pritiskom tipke «▼» napredujete v meni «Izbira komunikacijskega parametra» (Glejte poglavje 3.2.2).
3.2.4. Ponastav	Itev parametrov
SEL	S pritiskom tipke (=>) lahko uporabnik resetira različne sisteme parametrov (Glej poglavje 3.2.4.1)**.
PRrR	S pritiskom tipke «▲» se pomaknete v meni «Izbira izhodnih možnosti» (Glejte poglavje 3.2.5) in s pritiskom tipke «▼» napredujete v meni «Izbira komunikacijskih poznatorum (Ke postale komunikacijskih poznatorum)
SEŁuP	ethernet modula, napreduje na izbiro parametra IP, sicer gre na izbiro serijske komunikacija) (Glej oddelek 3.2.2)

** - Glej celotne navodila

3.2.5. Izbira izhodnih možnosti

Ta meni omogoča uporabniku da izbere izhodne možnosti



npr. «Relejni izhod». S pritiskom gumba « » se uporabniku omogoča da izbere izhodne možnosti (Glej poglavje 3.2.5.1)**.

S pritiskom gumba «▲» se uporabniku omogoča dostop v meni «Datalog Option Selection» (Glej poglavje 3.2.7) in s pritiskom tipke tipke «▼» v prejšnji meni (Glej poglavje 3.2.5).

3.2.6. Možnost izbire dnevnika podatkov

SEL drer Lou SEŁuP

5EL

SELUP

SEL rEC

SELUP

d ,SP PRrR

Ta meni omogoča uporabniku, da izbere zapis podatkov po opcija, npr. «Po dogodkih», «Po času» in «Po bremenu». S pritiskom gumba « » se uporabniku omogoča da izbere zapis oodatkov po možnostih (Glej poglavje 3.2.6.1)*; S pritiskom «A» se uporabniku omogoča dostop v meni «Parametri zaslona» (Glej poglavje 3.2.7) in s pritiskom tipke tipke «▼» v prejšnji meni (Glej poglavje 3.2.5).

3.2.7. Možnost nastavitve zaslona

Ta meni omogoča uporabniku, da izbere različne nastavitve zaslona, kot npr. «Osvetljenost ozadja», «Zaslon», «Kontrast» in «Stare nastavitve».

S pritiskom gumba « » se uporabniku omogoča, da izbere različne nastavitve (Glej poglavje 3.2.7.1)**. S pritiskom «A» se uporabniku omogoča vstop v meni «RTC 4.3. Ožičenje

nastavitve» (Glej poglavje 3.2.8) in s pritiskom gumba «▼» v prejšnji meni (Glej poglavje 3.2.6).

3.2.8. Nastavitve RTC

ĪΡ

Ta meni omogoča uporabniku, da izbere nastavitve kot so «Nastavitev datuma» in «Nastavitev ure».

S pritiskom gumba «) se uporabniku omogoča, da nastavi datum in uro (Glej poglavje 3.2.8.1)**. S pritiskom «A» se uporabniku omogoča vstop v meni «Tovarniške

3.2.9. Tovarniške nastavitve

SEL FREE rSEE SEŁuP

Ta meni omogoča uporabniku, da napravo ponastavi na Tovarniške nastavitve (Glej poglavje 3.2.9.1)**. S pritiskom gumba «a» se uporabniku omogoča preklop v «Konec nastavitev» (Glej poglavje 3.2.10) in s pritiskom «▼» se uporabnik vrne v meni «RTC nastavitve» (Glej poglavje 3.2.9).

3.2.10. Konec nastavitev

SEL 9....E PRrR SEŁuP

S pritiskom gumba «페» se uporabniku omogoča Konec nastavitvenega menija & vračanje v merilni zaslon. S pritiskom gumba «A» se uporabniku omogoča preklop v «Izbiro

Ta meni omogoča uporabniku, da konča nastavitveni meni (Glej poglavje 3.2.10.1)**.

sistemskih parametrov» (Glej poglavje 3.2.1) in s pritiskom «▼» se uporabnik vrne v meni «Tovarniške nastavitve» (Glej poglavje 329

** - Glej razširjena navodila/priročnik

4. VGRADNJA MERILNIKA

Vgradnja naprave je enostavna s "Cipin" nastavki. Vstavite napravo v odprtino (velikosti 92x92mm) in ko klikne se naprava prileže v štiri vgrajene zadrževalne sponke. Po potrebi je dodatna podpora na voljo z vrtljivim elementom, kot je prikazano na slik.

Sprednji del naprave ustreza razredu zaščite IP54. Za dodatno zaščito je mogoče uporabiti tesnilno ploščo. Kontakti na zadnji strani naprave so zaščiteni pred tekočinami.

Naprava naj bo vgrajena v stabilnem okolju kjer se temperatura okolice giblje med 0 - 50 °C. Vibracije naj bodo čim manj in naprava naj ne bo izpostavljena soncu



Pozor

v delovaniu

1. Zaradi varnosti in funkcionalnosti mora ta izdelek vgraditi usposobljena oseba in v skladu z lokalnimi predpisi.

2. Na nekaterih prikliučkih naprave so prisotne napetosti, nevarne za človeško živlienie. Pred poskusom kakršnekoli povezave ali odklopa je potrebno odklopiti napravo iz el. omrežja. 3. Naprava nima vgrajene varovalke, zato mora biti varovana z zunanjimi varovalkami.

4.1. Zahteve za vgradnjo/namestitev EMC

Naprava je zasnovana tako, da ob vgradnji v industrijsko okolje ustreza certifikatom in direktivam EU.

1. Prikazani izhodni in vhodni nizki signali imajo vgrajeno opremo za preprečevanje RF, kot so feritni absorberji, linijski filtri itd., V primeru, da RF polja povzročajo težave. Opomba: Priporočljivo je, da občutljive elektronske instrumente, ki opravljajo kritične meritve, namestite v ohišja EMĆ, ki jih ščitijo pred električnimi motnjami, ki bi lahko povzročile motnje

2. Izogibajte se napeljavi kablov/žic ob kablih in napravah ki so vir ali lahko povzročajo 6. VTIČNI MODUL motnie

3. Za zaščito izdelka pred trajnimi poškodbami je treba prehodne napetosti omejene na 2 kV. Dobra praksa je da se diferenčna napetost omeji na 2kV pri izvoru. Theunit je predviden za samodejno okrevanje v primeru visoke stopnje prehodnih razmer. V skrajnih primerih bo morda treba začasno odklopiti pomožno napajanje za več kot 5 sekund, da se ponovno vzpostavi pravilno delovanie.

Tokovni vhodi teh izdelkov so namenjeni samo za povezavo s sistemi prek tokovnih transformatorjev, kjer je ena stran ozemljena.

4. Pri ravnanju s tem izdelkom morate ves čas upoštevati varnostne ukrepe.

4.2. Dimenzija ohišja in izrez plošče



Vhodne povezave so izvedene neposredno na vijačne sponke s posrednim pritiskom žice. Oštevilčenje je jasno označeno na priključku. Izbira kabla mora ustrezati lokalnim predpisom. Priključne sponke za tok in napetost so do velikosti premera do 4 mm2 (12AWG) ali 2,5 mm2 fino-žičnega kabla

Note: It is recommended to use wire with lug for connection with meter.

4.4. Pomožno napajanje

Naprava mora biti napajana iz svojega izvora, ne glede na to, ali se napaja iz vira signala, nastavitve» (Glej poglavje 3.2.8) in s pritiskom gumba «v» v prejšnji meni (Glej poglavje 3.2.7).

4.5. Varovanie

Priporočljivo je, da so vsi napetostni vodi opremljeni z 1A HRC varovalko.

4.6. Ozemljevanje

Iz varnostnih razlogov, je potrebno sekundarne CT sponke priporočljivo ozemljiti skladno z lokalnimi predpisi in standardi.

5. VEZALNE SHEME



	Insert the Addon Assembly here and fit the assembly with the screws	
	Pluggable Module	
7. SPECIFI	ACIJE	
Sistem	3 taze 3/4 žični sistem ali enofaz programirljiv na mestu	ni
Vhodi		

Sistem	3 faze 3/4 zichi sistem ali enofazhi	onapria
	programirljiv na mestu	Debelina
Vhodi		Teža
Nazivne vhodne vrednosti (AC RMS)	57,7V LN do 346,4V LN programirliive na mestu	Izhodni Kotakti r
Sistem DT Drimerne vrednesti	100VLL do 1200kVLL	Priključr
Sistem PT Primarne vrednosti	programirano na mestu	Privzeti
	100V LL do 600V LL	
Sistem PT Sekundarne vrednosti	57,7V LN do 346,4V LN	
	programirljive na mestu	Delitev
MAXstalna napetost na vhodu	120% Nominalne vrednosti	10
Nazivna napetost vnoda		
NAZIVNI VNOGNI TOK	1200/ nozirma vradnosti	
Nazivni vhodni bromonski tok	120% lidzivile vieuliosu	100
	Standardne vrednosti 1-0000A (1 ali 5 Amn	
Sistem CT primarne vrednosti	na sekundariu)	1000
Sistem CT selvundernih vrednesti	1A/5A	1000
Sistem CT sekundarnin vrednosti	programirano na mestu	
Prikaz preobremenitve	"-OL-" >121%nazivne vrednosti (napetosti	Traianie
Traina proobromonitov	all toka)	· · j · · j -
frajna preobrementev	2 x oconiona vrodnost (1s anlikacija ob 10	Opomb
Napetostni vhod	intervalih) se ponovi 10x	1. Glej 1 2. Zgora
Trenutni vhod	intervalih) se ponovi 5x	MadDu
Dodatna pomožna oprema	/ 1	Drotoka
Zunanji višji pomožni	100V - 550V AC/DC	Hitrost
Zunanji višji pomožni nominalna vrednost	230V AC/DC 50 Hz za AC izhod	Thuosup
Obseg izhodne frekvence	45 - 65 Hz	Parnost
VA obremenitev s kartico Addon	<6VA približno (pri nominalni vrednosti)	Etherne
VA obremenitev s kartico Ethernet	<8VA približno (pri nominalni vrednosti)	Izhodni
Delovni merilna območja		Impulzn
Napetost	20 120 % nazivne vrednosti	
Tok	1 120 % nazivne vrednosti	
Zagonski tok		8 PO
D	po IEC 62053-22 (0,5 s / 0,2 s)	
Pogostost	45 - 66 HZ	RS48
Faktor moci	0,5 Zaostanek 1 0,8 prenitevanje	
Harmonično popačenje	10% do 31 harmonika	1 RS
Prinoročeni pogoji za natančnost delova	ania	1.1.0
Priporočena temperatura (po IEC 62053-21)	23°C ± 2°C	
Točnost		•
Napetost	±0,5 % vrednosti	
Tok	±0,5 % vrednosti	
Frekvenca	±0,2 % povprečne frekvence	
Delovna moč	±0,5 % vrednosti	
Jalova moč	±1,0 % vrednosti	Model
Navidezna moč	±0,5 % vrednosti	
Faktor moči/fazni kot	±3°	END25
Delovna moc	po IEC 62053-22 (0.5 s)	END25
lalava na s	po IEC 62053-22 (0.2 s) (opcija)	LIND25
Jalova moc	Class 2 po IEC 62053-23	
THD (Napetost/Tok)	±3,0 %	
Nonotost	+0.2.% vrodposti	
Tok	±0,2 % vrednosti	Celotna
Frekvenca	+0.2 % povprečne frekvence	
Delovna moč	+0.2 % vrednosti	
Navidezna moč	+0.2 % vrednosti	
Ekran		
LCD zaslon z osvetlienim ozadiem		
Reakcijski čas	Približno 1 s	
Kontrolne tipke		
Uporabniški vmesnik	3 tipke	
Standardi		
Imunost na EMC	IEC 61326 - 1 : 2012	
law and	IEC 61000-4-3, 10V/m min –	
intunost	3 stopnja industrijski nižji nivo	
Varnost	IEC 61010-1-2010	
	stalen priklop	
IP (azred	IEU 00529	





Okvir Dim. za Skupna ebelin eža hodni otakti rikliučr rivzeti

elitev

)pomb

arnos thern zhodni npulzna LED

В Model END25

Stopnja onesnaženosti	2
Kategorija inštalacije	III
Izolacija	
Razred zaščite	2
Test prenapetosti	
Vhod/izhod proti izvoru	4 kV RMS, 50 Hz, 1 min
Vhod proti ostalem tokokrogu	3,3 kV RMS, 50 Hz, 1 min
Okoljski pogoji	
Delovna temperatura	-10 do +60°C
Temperatura skladiščenja	-20 do +70°C
Relativna vlažnost	0 90% (brez kondenzacije)
Čas ogrevanja	3 min (minimalno)
Shock test (po IEC 60068-2-27)	Polovica sinusoide, vrh naraščanja, 30gn (300m/s2), v trajanju 18ms
Vibracije	1015010 Hz, 0,15mm amplitude
Št. ciklusov čiščenja	10 na os
Stopnja zaščite	
Zaščita spreda	IP54
Zaščita zadaj (priključne sponke)	IP20
Dimenzije	
Okvir	96 x 96 mm DIN 43718
Dim. za izrez	92 mm +0,8 x 92 mm +0,8
Skupna globina	75 mm
Debelina panela	1 - 3 mm za samoklikanje 1 - 6 mm za vijake
Teža	320 za Approx
Izhodni impulzi Opcije	
Kotakti releja	1 NO
Priključna napetost&tok	5 A, 240 V AC
Privzeti delitelj impulza	1 na Wh (do 3600W)
	1 na kWh (do 3600kW)
	1 na MWh (med 3600 do 30 000kW)
Delitev impulzne frekvence	Nastavi se na mestu
10	1 za 10Wh (do 3600W)
	1 za 10kWh (do 3600W)
	1 za 10MWh (med 3600W do 30 000kW)
100	1 za 100Wh (do 3600W)
	1 za 100kWh (do 3600W)
	1 za 100MWh (med 3600W do 30 000kW)
1000	1 za 1000Wh (do 3600W)
	1 za 1000kWh (do 3600W)
	1 za 1000MWh (med 3600W do 30 000kW)
Trajanje impulza	60ms, 100ms ali 200ms
Opombe: 1. Glej Tabelo 4 za podrobnosti**.	

Zgoraj navedeni pogoji veljajo tudi za jalovo in navidezno energijo

lodBus (RS485) Možnost

- (
1	ModBus (RS485)
orenosa	4.8k, 9.6k, 19.2k, 38.4k, 57.6kbps
:	Nenavadno ali sodo, z enim zapornim bitom, ali nobena z 1 ali 2 stop bitoma
et	Ethernet dostop na Modbus TCP/IP Protokol
impulz	
a LED	Za preizkušanje energije

POVEZAVA NA OPTIČNI IMPULNI IZHOD / S485 (POGLED OD ZADAJ NA UNIVEZALNI INŠTRUMENT)

RS 485 izhod z Relejem 1 & Relejem 2

••	۲°	 。	- С о	
A Gnd	NO	COM	NO	COM
RS 485		ay 1	Rel	ay 2

Ključne funkcije

RS	RS485interface + 2pulzna izhoda, razred točnosti 0,5s
ETH	Ethernet + beleženje podatkov, razred točnosti 0,2s

elotna navodila so dostopna tu www.etigroup.eu in s kodo QR







Tel. : +386 03 56 57 570 e-mail: eti@eti.si



POWER NETWORK METER



USER'S MANUAL

Installation & Operating Instructions

Section	Contents
1.	Introduction
2.	Measurement and Energy/Counter Screens
	2.1 Current Reversal Screen
	2.2 Phase Rotation Error Screen
	2.3 Phase Absent Screen
	2.4 Real Time Clock
	2.5 Individual Harmonics
	2.6 Timer 1 & Timer 2 Screens
	2.7 Run Hour
	2.8 On Hour
	2.9 Number of Interruption
3.	Programming
	3.1 Password Protection
	3.2 Menu Selection
	3.2.1 System Parameter Selection
	3.2.1.1 System Type
	3.2.1.2 Potential Transformer Primary Value
	3.2.1.3 Potential Transformer Secondary Value
	3.2.1.4 Current Transformer Primary Value
	3.2.1.5 Current Transformer Secondary Value
	3.2.1.6 System Frequency
	3.2.1.7 Demand Integration Time
	3.2.1.8 Auto Scrolling
	3.2.1.9 Low Current Noise Cutoff
	3.2.1.10 No. of Poles
	3.2.1.11 Energy Output
	3.2.1.12 Energy Digit Reset Count
	3.2.1.13 Energy Rate
	3.2.1.14 Impulse Rate
	3.2.1.15 Quit System Parameters

- 3.2.2 Communication Parameter Selection
 - 3.2.2.1 Address Setting
 - 3.2.2.2 RS 485 Baud Rate
 - 3.2.2.3 RS 485 Parity
 - 3.2.2.4 Quit Communication Parameters
- 3.2.3 IP Parameter Selection
 - 3.2.3.1 IP Address Setting
 - 3.2.3.2 Subnet mask Setting
 - 3.2.3.3 Default gateway Setting
 - 3.2.3.4 Server Port Setting
 - 3.2.3.5 Quit IP Parameters
- 3.2.4 Reset Parameter Selection
 - 3.2.4.1 Resetting Parameter
- 3.2.5 Output Option Selection
 - 3.2.5.1 Relay Selection
 - 3.2.5.1.1 Relay 1 or 2 Output Selection Menu
 - 3.2.5.1.1.1 Pulse Output
 - 3.2.5.1.1.1.1 Parameter Selection
 - 3.2.5.1.1.1.2 Pulse Duration
 - 3.2.5.1.1.1.3 Pulse Rate
 - 3.2.5.1.1.1.4 Quit Pulse Output
 - 3.2.5.1.1.2 Limit Output
 - 3.2.5.1.1.2.1 Limit Output Parameters
 - 3.2.5.1.1.2.2 Energy Count Configuration
 - 3.2.5.1.1.2.3 Energy Trip Point
 - 3.2.5.1.1.2.4 Energy Count ON Delay
 - 3.2.5.1.1.2.5 Parameter Configuration
 - 3.2.5.1.1.2.6 Trip Point
 - 3.2.5.1.1.2.7 Hysteresis
 - 3.2.5.1.1.2.8 Energizing Delay
 - 3.2.5.1.1.2.9 De-energizing Delay
 - 3.2.5.1.1.2.10 Quit Limit Output
 - 3.2.5.1.1.3 Timer
 - 3.2.5.1.1.3.1 Number of Cycles
 - 3.2.5.1.1.3.2 Timer Configuration
 - 3.2.5.1.1.3.3 On Delay
 - 3.2.5.1.1.3.4 Off Delay

3.2.5.1.1.3.5 Quit Timer Output

3.2.5.1.1.4 RTC Relay

3.2.5.1.1.4.1 Weekdays Selection

3.2.5.1.1.4.2 Relay Configuration

3.2.5.1.1.4.3 ON Time

3.2.5.1.1.4.4 OFF Time

3.2.5.1.1.4.5 Quit RTC Output

3.2.6 Datalog Option Selection

3.2.6.1 Event Based Datalog Setup

3.2.6.1.1 Event Based Datalog Selection

3.2.6.1.2 Quit Event Datalog

3.2.6.2 Time Based Datalog Setup

3.2.6.2.1 Time Based Datalog Selection

3.2.6.2.2 Time Interval Selection

3.2.6.2.3 Parameter Count

3.2.6.2.4 Parameter Selection

3.2.6.2.5 Quit Time Based Datalog

3.2.6.3 Load Profile Datalog Setup

3.2.6.3.1 Load Profile Datalog Selection

3.2.6.3.2 Quit Load Profile Datalog selection

3.2.6.4 Quit Datalog Option

3.2.7 Display Parameters

3.2.7.1 Feature Selection Menu

3.2.7.1.1 Backlit

3.2.7.1.2 User Assignable Screens

3.2.7.1.2.1 Screen Number Selection

3.2.7.1.2.2 User Screens Selection

3.2.7.1.2.3 Quit Userscreens

3.2.7.1.3 Contrast

3.2.7.1.4 Quit Display Parameters

3.2.8 RTC Setting Screen

3.2.8.1 Date Settings

3.2.8.2 Time Settings

3.2.8.3 Quit RTC

3.2.9 Factory Reset Screen

3.2.10 Quit

4.

Energy Auto-Ranging On Display

4.1 Calculating Display energy (Auto-ranging)

	4.2 Entering Energy start count
5.	Relay Output
	5.1 Pulse Output
	5.2 Limit Switch
	5.3 Timer Output
6.	Phasor Diagram
7.	Installation
	7.1 EMC Installation Requirements
	7.2 Case Dimensions and Panel Cut-out
	7.3 Wiring
	7.4 Auxiliary Supply
	7.5 Fusing
	7.6 Earth / Ground Connections
8.	Connection Diagrams
9.	Optional Pluggable Module
10.	Specification
11.	Connection for Optional Pulse Output / RS 485

1. INTRODUCTION

The Multifunction Instrument is a panel mounted 96 x 96mm DIN Quadratic Digital Panel Meter, which measures important electrical parameters in 3 ph 4 wire / 3 wire / 1 ph Network and replaces the multiple analog panel meters. It measures electrical parameters like AC voltage, Current, Frequency, Power, Energy (Active / Reactive / Apparent), phase angle, power factor, individual harmonics & many more. The instrument integrates accurate measurement technology (All Voltages & current measurements are True RMS upto 31st Harmonic) with LCD display with backlit.

It can be configured & Programmed at site for the following : PT Primary, PT Secondary, CT Primary, CT Secondary 3 Phase 3W, 3 Phase 4W, 1 Phase 2W system.

The front panel has three push buttons using which the user can scroll through different screens & configure the product. The front panel also has Impulse red led, flashing at rate proportional to measured power.



2. MEASUREMENT AND ENERGY/COUNTER SCREENS

In normal operation, the user is presented with two simultaneous screens:

1. One of the Measurement screens out of the screens 1 to 36 of Table 1.

These screens may be scrolled through one at a time in incremental order by pressing the "Up key" and in decremental order by pressing "Down key". Few important screens are explained in Section 2.1 to 2.6.

2. One of the Energy/Counter screens out of the screens 37 to 62 of Table 1.

These screens may be scrolled through one at a time in incremental order only by pressing the "Enter key" to roll over again in the same order. Few important screens are explained in Section 2.7 to 2.9.

Load Graphics

Load Graphics indicates the input current as the percentage of the CT Primary value. This indication is available for all measurement screens. For example, consider CT Primary to be set at 5A, then the input current of 2.5 A indicates 50% as shown below.



The absence of lines indicating the percentage implies that the input current is less than 20% of the CT Primary value.

Phase Sequence Indication

It indicates the rotation of input phasor vectors :

clockwise/ counter-clockwise.

In case the input is absent or the phase sequence is neither L123 nor L321, the phase sequence indication is not shown.



Clockwise Sequence for L123.



Counter-Clockwise Sequence, for L321

Screen	Parameters		On Display			On Modbus		
No.	r arameters	3P 4W	3P 3W	1P 2W	3P 4W	3P 3W	1P 2W	
1	System Voltage/ Current/ Power/ Frequency	√	✓	✓	✓	✓	✓	
2	L-N Voltage	 ✓ 	×	×	✓	×	×	
3	L-L Voltage	✓	✓	×	✓	✓	×	
4	Current ,Neutral Current	 ✓ 	√*	×	✓	√*	×	
5	Phase Voltage THD	×	×	×	×	×	×	
6	Phase Current THD	×	x	×	×	×	×	
7	L1 phase VA/VAr/W	 ✓ 	x	×	✓	×	×	
8	L2 phase VA/VAr/W	✓	×	×	✓	×	×	
9	L3 phase VA/VAr/W	 ✓ 	×	×	✓	×	×	
10	Phase Angle	×	x	×	×	×	×	
11	VA/A demand	×	×	×	×	×	×	
12	Capacitive, Inductive VAr demand	×	×	×	×	×	×	
13	W IMP demand	×	×	×	×	×	×	
14	W EXP demand	×	x	×	×	×	×	
15	Max VA/A demand	×	×	×	×	×	×	
16	Max Capacitive, Inductive VAr demand	×	×	×	×	×	×	
17	Max W IMP demand	×	x	×	×	×	×	
18	Max W EXP demand	×	×	×	×	×	×	
19	Old Max VA/A demand	×	×	×	×	×	×	
20	Old Max Capacitive, Inductive VAr demand	×	×	×	×	×	×	
21	Old Max W IMP demand	×	×	×	×	×	×	
22	Old Max W EXP demand	×	x	×	×	×	×	
23	Sys RPM/Frequency	✓	\checkmark	√	✓	✓	✓	
24	Sys Active/Reactive/Apparent power	✓	✓	√	✓	✓	✓	
25	Sys Apparent, Reactive (Power), Phase angle, PF	√	\checkmark	✓	✓	✓	✓	
26	Min Sys Voltage / Current	✓	✓	√	✓	✓	✓	
27	Max Sys Voltage / Current	✓	\checkmark	√	✓	✓	✓	
28	Sys %THD Voltage /Current	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	
29	Current Reversal	✓	×	\checkmark	×	x	x	
30	Phase rotation error	\checkmark	\checkmark	×	\checkmark	\checkmark	×	
31	Phase absent indication	 ✓ 	×	×	×	x	×	
32	RTC	✓	✓	√	\checkmark	✓	✓	

TABLE 1 : Measurement & Energy/Counter Screens for Basic version of END25

*Note: In 3P3W system, Neutral Current is not shown, only line currents are shown.

TABLE 1 : Continued...

Screen	Parameters		On Display			On Modbus		
No.			3P 3W	1P 2W	3P 4W	3P 3W	1P 2W	
33	Individual harmonics V	×	×	×	×	×	×	
34	Individual harmonics A	×	×	×	×	×	×	
35	Timer1 No. of Cycles, ON, OFF delay	√	✓	✓	✓	✓	✓	
36	Timer2 No. of Cycles, ON, OFF delay	√	✓	✓	✓	✓	✓	
37	Active Energy Import(Overflow)	×	×	×	✓	✓	~	
38	Active Energy Import	✓	✓	\checkmark	✓	✓	✓	
39	Active Energy Export(Overflow)	×	×	×	 ✓ 	✓	✓	
40	Active Energy Export	√	✓	✓	 ✓ 	✓	✓	
41	Reactive Capacitive energy (Overflow)	×	×	×	✓	✓	✓	
42	Reactive Capacitive energy	✓	✓	✓	✓	✓	~	
43	Reactive Inductive energy (Overflow)	×	×	×	✓	✓	✓	
44	Reactive Inductive energy	✓	✓	✓	✓	✓	✓	
45	Apparent energy(Overflow)	×	×	×	✓	✓	✓	
46	Apparent energy	√	✓	✓	✓	✓	✓	
47	Run hour	✓	✓	✓	 ✓ 	 ✓ 	✓	
48	On hour	✓	✓	\checkmark	✓	✓	\checkmark	
49	No. of interrupts	✓	\checkmark	\checkmark	 ✓ 	✓	\checkmark	
50	Old Active energy Import(Overflow)	×	×	×	×	×	×	
51	Old Active energy Import	×	×	×	×	×	×	
52	Old Active energy Export(Overflow)	×	×	×	×	×	×	
53	Old Active energy Export	×	×	×	×	×	×	
54	Old Reactive Capacitive energy (Overflow)	×	×	×	×	×	×	
55	Old Reactive Capacitive energy	×	×	×	×	×	×	
56	Old Reactive Inductive energy (Overflow)	×	×	×	×	×	×	
57	Old Reactive Inductive energy	×	×	×	×	×	×	
58	Old Apparent energy(Overflow)	×	×	x	×	×	×	
59	Old Apparent energy	×	×	x	×	×	×	
60	Old Run hour	×	×	x	×	x	×	
61	Old On hour	×	×	×	×	×	×	
62	Old no. of interrupts	×	×	x	×	x	×	

Note: (1) Only screens (with screen number) 1 to 32 are available for selectable Userscreens.

(2) Energy on Display is Auto-Ranging For details refer Section 4.1.

2.1 Current Reversal Screen

This screen is useful to indicate if current in any phase is reversed or not. If current in any phase gets reversed, then corresponding phase will be indicated on this screen.



This screen shows that currents in L1 and L3 are reversed.



This screen shows that currents in all three phase are correct.



This screen shows that the meter has no current input.

2.2 Phase Rotation Error Screen

Meter shows phase rotation information for the phase sequence R-Y-B (L1-L2-L3).



User must check this screen in order to get correct readings when meter is connected.

2.3 Phase Absent Screen

This screen is useful to indicate if voltage or current in any phase is absent. Hence, user will know which voltage or current is missing and take corrective action.



This screen indicates that all three phases (volt. & current) are absent.

L Z Ž

This screen indicates that V2, I2 and I3 are absent.



This screen indicates that all three phases are present i.e. all inputs are present.

2.4 Real Time Clock



2.5 Individual Harmonics



The **Individual Harmonics** can be accessed by pressing the Enter key followed by the Up and Down keys taking through the 31 harmonics.



Similarly, pressing the " - " key at the **quit screen** takes the user out of the Individual Harmonics Screen.

2.6 Timer 1 & Timer 2 Screens



The screen shows the No. of Cycles, on delay, off delay of the corresponding relay in its timer mode. If the relay is configured in timer mode, then the timer can be turned ON by long press (about 3sec) of Up key while present in any of the measurement screens.



Configuration.



5EL Eur 1 гилл

Relay 1 is not selected in Timer This screen is shown when relay is configured as timer.by selecting yes option timer1 can be started.

Timer showing Running when it is started.

Similarly, the Relay can be turned OFF by long press (about 3sec) of Down key while present in any of the measurement screens.



Relay 1 is not selected in Timer Configuration.



This screen is shown when timer is started .by selecting yes option timer1 can be stopped.



Timer showing Stopped when not running.

Similarly, for Timer Configuration, Relay 2 parameters and status can also be set and accessed (respectively) in a way similar to that of Relay 1.

2.7 Run Hour



2.8 On Hour



This Screen shows the total no. of hours the load is connected. Even if the Auxiliary supply is interrupted, count of Run hour will be maintained in internal memory & displayed in the format "hours. min".

For example if Displayed count is 105000.10 it indicates 105000 hours & 10 minutes. After 999999.59 run hours display will restart from zero. To reset run hour manually see section Resetting Parameter 3.2.3.1.

This Screen shows the total no. of hours the Auxiliary Supply is ON. Even if the Auxiliary supply is interrupted count of On hour will be maintained in internal memory & displayed in the format "hours. min".

For example if Displayed count is 105000.10 it indicates 105000 hours and 10 minutes. After 999999.59 On hours display will restart from zero. To reset On hour manually see section Resetting Parameter 3.2.3.1.

2.9 Number of Interruption



This Screen Displays the total no. of times the Axillary Supply was Interrupted. Even if the Auxiliary supply is interrupted count will be maintained in internal memory. To reset No of Interruption manually see section Resetting Parameter 3.2.3.1.





3. PROGRAMMING

The following sections comprise step by step procedures for configuring the Multifunction Meter according to individual user requirements. To access the set-up screens press and hold " \blacktriangle UP" and " \checkmark DOWN" keys simultaneously for 5 seconds. This will take the User into the Password Protection Entry Stage (Section 3.1).

3.1. Password Protection

Password protection can be enabled to prevent unauthorised access to set-up screens, when default password protection is not enabled. Password protection is enabled by selecting a four digit number other than 0000, esting a password of 0000 disables the password protection.





donE

Enter Password, second digit entered, prompt for third digit,

Press the "A" key to scroll the value of first digit from 0 through to 9, the value rolls back from 9 round to 0 and " V " key to scroll the value of first digit from 9 through to 0, the value rolls back from 0 round to 9.

Press the "-" key to advance to next digit.

Enter Password, third digit entered, prompt for fourth digit. .

Press the " A " key to scroll the value of first digit from 0 through to 9, the value rolls back from 9 round to 0 and " **v** " key to scroll the value of first digit from 9 through to 0, the value rolls back from 0 round to 9.

Press the "" key to advance to verification of the password.

Enter Password, fourth digit entered, awaiting verification of the password.

Password Incorrect.



The unit has not accepted the Password entered.

Pressing the "▲" or " ▼ " key advances to the Enter Password stage.

Pressing the " \blacksquare " key exits the Password menu & returns operation to the measurement reading mode.

New / Change Password

EndE

0000

ደሻ יF

FodF

2000 Ed.ik 

Press the "▲" and " ▼ " keys to scroll the value of first digit from 0 through to 9 and from 9 through to 0, respectively with digit roll around feature.

Pressing the "-" key advances the operation to the next digit and sets the first digit, in this case to "2".





Press the " \blacktriangle " and " \blacktriangledown " keys to scroll the value of second digit from 0 through to 9 and from 9 through to 0,respectively with digit roll around feature.

Pressing the "_" key advances the operation to the next digit and sets the second digit, in this case to "1".



New/ Change Password, second digit entered, prompting for third digit.

Press the " \blacktriangle " and " \blacktriangledown " keys to scroll the value of second digit from 0 through to 9 and from 9 through to 0,respectively with digit roll around feature.

Pressing the "- " key to advance the operation to the next digit and sets the third digit, in this case to "5".



New/ Change Password, third digit entered, prompting for fourth digit. .

Press the " \blacktriangle " and " \blacktriangledown " keys to scroll the value of second digit from 0 through to 9 and from 9 through to 0,respectively with digit roll around feature.

Pressing the "—" key to advance the "New Password Confirmed" and sets the fourth digit, in this case to "3".

New Password confirmed



Pressing the " \blacktriangle " or " \blacktriangledown " key returns to the "New/Change Password" stage.

Pressing the " - " key advances to the Menu selection screen (see Section 3.2).

3.2 Menu selection.

3.2.1 System Parameter Selection



This screen is used to select different system Parameters like "system type", "CT Ratio", "PT Ratio", etc. Pressing the " = " key allows the user to set Different system parameters (see Section 3.2.1.1 to 3.2.1.12).

Pressing the "▲" key advances to the "Communication Parameter Selection" screen (see section 3.2.2) and pressing "▼" key advances to the "Quit Setup" Screen (see section 3.2.9).

3.2.2 Communication Parameter Selection



This screen is used to select the different communication parameters like "Address selection", "RS485 Parity selection", "RS485 baud rate", etc.

Pressing the " = " key allows the user to set different Communication parameters (see Section 3.2.2.1 to 3.2.2.3).

Pressing the "▲" key advances to the "Reset Parameter Selection" screen (if ethernet module is present then it will advance to IP parameter selection see section 3.2.3) (see section 3.2.4) and pressing "♥" key advances to the "System Parameter Selection" screen (see Section 3.2.1).

3.2.3 IP Parameter Selection



This screen is used to select the different communication parameters like "IP Address selection", "Subnet mask selection", "default gateway selection", server port selection etc.

Pressing the "—" key allows the user to set different Ethernet parameters (see Section 3,2,3,1 to 3,2,3,5).

Pressing the "▲" key advances to the "Reset Parameter Selection" screen (see section 3,2,4) and pressing "▼" key advances to the "Communication Parameter Selection" screen (see Section 3,2,2).

3.2.4 Reset Parameter Selection



This screen is used to reset different parameters.

Pressing the " - " key allows the user to reset different system parameters (see Section 3.2.4.1).

Pressing the *▲ * key advances to the "Output Option Selection" screen (see Section 3.2.5) and pressing * ▼ * key advances to "Communication Parameter Selection" screen(ethermet module is present then it advances to IP parameter selection else it goes to serial communication parameter selection) (see Section 3.2.2).

3.2.5 Output Option Selection



This screen will allow the user to select Output Options like "Relay Output".

Pressing the " - " key allows the user to select and configure the output options (see Section 3.2.5.1).

Pressing "▲" key advances to the "Datalog Option Selection" screen (see section 3.2.6) and pressing "▼ " key advances to "Reset Parameter Selection" screen (see Section 3.2.4).

3.2.6 Datalog Option Selection



This screen will allow the user to select Datalog Options like "Event Based", "Time Based" and "Load Profile".

Pressing the " — " key allows the user to select and configure the datalog options (see section 3.2.6.1).

Pressing the "▲ " key advances to "Display Parameters" screen (see section 3.2.7) and pressing "▼" key advances to "Output Option Selection" screen (see Section 3.2.5).

3.2.7 Display Parameters



This screen will allow the user to access different features like "Backlit", "Screens", "Contrast" and "Old Parameters".

Pressing the " - " key allows the user to select and configure the features (see Section 3.2.7.1).

Pressing the "▲" key advances to "RTC Setting" screen (see Section 3.2.8) and pressing "▼ " key advances to "Datalog Option Selection" screen (see Section 3.2.6).

3.2.8 RTC Setting



This screen will allow the user to access features like "Set Date" and "Set Time".

Pressing the "- " key allows the user to select date and time (see Section 3.2.8.1).

Pressing the "▲" key advances to "Factory Reset" screen (see Section 3.2.9) and pressing "▼" key advances to "Display Parameters" screen (see Section 3.2.7).

3.2.9 Factory Reset



This screen allows the user to set the meter to its Factory Default settings (see Section 3.2.9.1).

Pressing the "▲" key advances to "Quit Setup"screen (see Section 3.2.10) and pressing " ▼ " key advances to "RTC Setting" screen (see Section 3.2.9).

3.2.10 Quit Setup



This screen will allow the user to quit the setup menu(see Section 3.2.10.1)

Pressing the " - " key allows the user to Quit from setup menu & return to measurement screen.

Pressing the "▲" key advances to "System Parameter Selection" screen (see Section 3.2.1) and pressing "▼ " key advances to "Factory Reset" screen (see Section 3.2.9).

3.2.1 System Parameter Selection 3.2.1.1 System Type



This screen is used to set the system type.

Pressing the "-" key advances into the system type edit mode and pressing the " \blacktriangle " and " \checkmark " key scrolls through the options available:

System type "3" for 3 phase 3 wire, "4" for 3 phase 4 wire system & "1" for single phase system.

Pressing the "_"'key advances to "PIs Wait" screen, accepting the present value and returns to the "System Type" menu (see Section 3.2.1.1).



Once the System type is set and "System Type" menu appears (see Section 3.2.1.1), pressing " \triangleq " key advances to the "Potential Transformer Primary Value" screen (see Section 3.2.1.2) and pressing the " \forall key advances to "Quit System Parameters" screen (see Section 3.2.1.14). Default value is set to 4" i.e. 3P 40%.

1 Caution: If System type is changed then log of data will reset.

3.2.1.2 Potential Transformer Primary Value

The nominal full scale voltage is displayed as the Line to Line voltages for all system types. The values displayed represent the voltage in kilovolts (note "k" symbol).



Pressing the "▲" key accepts the present value and advances to the "Potential Transformer Secondary Value" screen (see Section 3.2.1.3).

Similarly, pressing the " ▼ " key accepts the present value and advances to the "System Type" screen (see Section 3.2.1.1).

Pressing the " - " key advances to the "Potential Transformer Primary Decimal Point Edit" mode.

Potential Transformer Primary Decimal Point Edit



(Flashing decimal point indicates the cursor position).Initially the "decimal point must be selected, pressing the "h" or " Ψ " key moves the decimal point position to the right until it disappears, which means that it has reached ###. after which it returns to #.###.

Note: The absence of decimal point in edit mode implies # # # #. decimal point position.

Pressing the " - " key accepts the present decimal point position and advances to the "Potential Transformer Primary Digit Edit" mode.

Potential Transformer Primary Digit Edit



Pressing the "▲" or "▼ key scrolls the value of the most significant digit from 0 through 9 or 9 through to 0, respectively unless the present displayed Potential Transformer Primary Value together with the Current Transformer Primary Value, previously set, results in a maximum system power of greater than 5400 MVA (1800 MVA per phase) in which case the digit range gets restricted.

Pressing the "- " key accepts the present value at the cursor position and advances the cursor to the next less significant digit.

The PT Primary value can be set from 100 VL- L to 1200 kVL-L. The value will be forced to 100 VL-L if set less than 100.

Note: The flashing digit indicates the cursor position, a steady decimal point is present to identify the scaling of the number until the cursor position coincides with the steady decimal point position. At this stage the digit will flash.

When the least significant digit has been set, pressing the "- " key shows "PIs Wait" screen which is followed by the "Potential Transformer Primary Value" screen (see Section 3.2.1.2).

Note : PT Values must be set as Line to Line Voltage for Primary as Well as Secondary for all system types (3P3W/3P4W/1P2W).

The default value is '0.415 kVLL'.

3.2.1.3 Potential Transformer Secondary Value

The value must be set to the nominal full scale secondary voltage which will be obtained from the Transformer when the potential transformer (PT) primary is supplied with the voltage defined in 3.2.1.2 Potential Transformer Primary voltage.The ratio of full scale primary to full scale secondary is defined as the transformer ratio.The PT Secondary value can be set from 100VL-L to 600VL-L (according to input voltage range).



Pressing the "▲" key accepts the present value and advances to the "Current Transformer Primary Value" screen (see Section 3.2.1.4).

Similarly, pressing the "▼ " key accepts the present value and advances to the " Potential Transformer Primary Value" screen (see Section 3.2.1.2). Pressing the " = " key advances to the "Potential Transformer Secondary Digit Edit" mode.

Potential Transformer Secondary Digit Edit



Pressing the "▲" or "▼" key scrolls the value of the most significant digit from 0 through 9 or 9 through 0, respectively.

Pressing the "- " key accepts the present value at the cursor position and advances the cursor to the next less significant digit.

After entering the least significant digit, pressing the " — " key sets the value and advances to the "PIs Wait" screen followed by the "Potential Transformer Secondary Value" screen (see Section 3.2.1.3). The default value is 415 VLL.

3.2.1.4 Current Transformer Primary Value

The nominal Full Scale Current that will be displayed as the Line currents. This screen enables the user to display the Line currents inclusive of any transformer ratios, the values displayed represent the Current in Amps.



Pressing the "▲" key accepts the present value and advances to the "Current Transformer Secondary Value" screen (see Section 3.2.1.5).

Similarly, pressing the "▼" key accepts the present value and advances to the "Potential Transformer Secondary Value" menu (see Section 3.2.1.3).

Pressing the " \blacksquare " key advances to the "Current Transformer Primary Digit Edit" mode.

Current Transformer Primary Digit Edit



Pressing the "▲" or "▼" key scrolls the value of the most significant digit from 0 through 9 or 9 through 0, respectively (with digit roll over feature) unless the present displayed Current Transformer Primary Value together with the Potential Transformer Primary Value results in a maximum system power of greater than 5400 MVA (1800 MVA per phase) in which case the digit range gets restricted, the value will wrap.

Example: If primary value of PT is set as 1200 kVL-L (max value) then primary value of Current is restricted to 1804 A.

Pressing the " - " key accepts the present value at the cursor position and advances the cursor to the next less significant digit.

The "Maximum Power" restriction of 5400 MVA refers to 200% of nominal current and 120% of nominal voltage, i.e, 2250 MVA nominal power .

After entering the least significant digit, pressing the " - " key sets the value and advances to the "PIs Wait" screen followed by "Current Transformer Primary Value" screen (see Section 3.2.1.4). NOTE: Default value is set to 5'i.e. 5A.

3.2.1.5 Current Transformer Secondary Value



This screen is used to set the secondary value for Current Transformer. Secondary value "5" for **5A or** "1" for **1A** can be selected.

Pressing the " **A** " key accepts the present value and advances to the "System Frequency" menu (see Section 3.2.1.6).

Similarly, pressing the "▼" key accepts the present value and advances to the "Current Transformer Primary Value" screen (see Section 3.2.1.4).

Pressing the " — " key advances to the "CT Secondary Value Edit" mode and keys "▲" and "▼" scroll the value through the options available.

Pressing the " - " key sets the option selected and advances to "PIs Wait" screen followed by "Current Transformer Secondary Value" screen (see Section 3.2.1.5).

NOTE: Default value is set to '5' i.e.

3.2.1.6 System Frequency



This screen is used to set the frequency of the input. The Unit of displayed values is $\ensuremath{\mathsf{Hz}}$.

Pressing the " ■ " key enables editing and pressing the " ▲ " or " ▼ " key scrolls through the following Options: **50,60** Hz.

Once the desired option has been selected, pressing " - " key confirms the selection and advances to the "System Frequency" menu (see Section 3.2.1.6).

Pressing the "▲ " key advances to "Demand Integration Time" screen (see Section 3.2.1.7) and pressing the "▼" key advances to "Current Transformer Secondary Value" screen (see Section 3.2.1.5).

Note: (1) The applied frequency and the entered frequency value should be same.

(2) Default value is set to '50' Hz.

3.2.1.7 Demand Integration Time

This screen is used to set the period over which current and power readings are to be integrated. The Unit of displayed values is **minutes**.



Pressing the " — " key enables editing and pressing keys " ▲ " and " ▼ " allows scrolling through the following Options: 8,15,20,30.

Once the desired option is selected, pressing " — " key confirms the selection and advances to "PIs Wait" screen followed by "Demand Integration Time" screen (see Section 3.2.1.7).

Pressing the "▲" key advances to "Auto Scrolling" screen (seeSection 3.2.1.8) and pressing the "▼" key advances to "System Frequency" screen (see Section 3.2.1.6).

NOTE: Default value is set to '8' i.e. 8 min.

3.2.1.8 Auto Scrolling



This screen allows user to enable screen scrolling.

Pressing "▲" key accepts the present status and advance to the "Low Current Noise Cutoff" screen (see Section 3.2.1.9). Similarly, pressing "▼" key accepts the present status and advances to the "Demand Integration Time" screen (see Section 3.2.1.7).

Pressing the " — " key allows editing and keys " \blacktriangle " and " \blacktriangledown " allows the user to select either 'Yes' to enable autoscroll and 'No' to disable autoscroll.

Pressing " = "key selects the status displayed and advances to "Auto Scrolling" screen (see Section 3.2.1.8). NOTE: (1) Default value is set to 'NO'.

(2) With Autoscrolling mode ON, Individual harmonics screen s are not displayed.

3.2.1.9 Low Current Noise Cutoff

This screen allows the user to set Low noise current cutoff in mA.



Pressing "▲" key accepts the present value and advance to "No. of Poles" screen (see Section 3.2.1.10). Similarly, pressing "▼ key accepts the present value and advance to "Auto Scrolling" screen (see Section 3.2.1.8).

Pressing the " \blacksquare " key allows editing and the user can select either 0 mA or 30 mA using " \blacktriangle " and " \blacktriangledown " keys.

Pressing the "..." key confirms the selection and advances to "Low Current Noise Cutoff" screen (see Section 3.2.1.9). NOTE: Default value is set as '0' mA.

3.2.1.10 No. of Poles

This screen enables to set No. of poles of a Generator of which RPM is to be measured and to which the instrument is connected to monitor its parameters.



Pressing "▲" key accepts the present selection and advances to "Energy Output" menu (see Section 3.2.1.11). Similarly, pressing "▼" key accepts the present selection and advances to "Low Current Noise Cutoff" screen (See section 3.2.1.9).

Pressing the " → " key advances to editing mode for no. of poles and pressing " ▲ " and " ▼ " keys scrolls the number from 2 to 40 and 40 to 2, respectively in steps of 2.

Pressing the "-" key selects the status displayed and enter the "No. of Poles" menu (see Section 3.2.1.10).

NOTE: Default value is set to '2'.

3.2.1.11. Energy Output



This screen enables user to set energy on modbus in terms of Wh / kWh / MWh as per the requirement . Same is applicable to all types of energy. Pressing " \blacktriangle " key accepts the presents value and advances to the "Energy Digit Reset Count" screen (see Section 3.2.1.12).

Similarly, pressing " ▼ " key accepts the present value and advances to the "No. of Poles" menu (see Section 3.2.1.10).

Pressing the "-" key will enter the editing mode for energy output and " ▲ " and " ▼ " keys scrolls through the values 1,2 & 3 and in the reverse order, respectively, with roll over feature:

- 1 : Energy In Wh
- 2 : Energy in kWh
- 3: Energy in MWh

Pressing the "-" key sets the value selected and advances to "PIs Wait" screen followed by "Energy Output" menu (see Section 3.2.1.11).

NOTE: 1. Default value is set to '2' i.e. Energy will be in terms of kWh/kVArh/kVAh respectively.

 If (PT primary(VLL) * CT primary * Root3) >30000 kW, then Energy Output can be set only as kWh and MWh.

- 3. Energy on display is auto-ranging & so this setting is only applicable for modbus energy counters.
 - (For Energy auto-ranging refer section on 4.1.)

3.2.1.12 Energy Digit Reset Count



This screen enables user for setting maximum energy count after which energy on modbus will roll over to zero. User can select one of: **7,8,9**.

Pressing the " \bigstar " key accepts the present value and will advance to the "Energy Rate" screen (see Section 3.2.1.13). Similarly, pressing the " \checkmark " key accepts the present value and will advance to the "Energy Output" menu (see Section 3.2.1.11).

Ex. If Energy Digit count is set to 9 then energy on modbus will reset after "999,999,999" & rollback to zero and simultaneously the corresponding **Overflow count** value increases by 1.

Pressing " - " key sets the value selected and advances to "PIs Wait" screen followed by "Energy Digit Reset Count" screen (see Section 3.2.1.12).

Note:(1) Default value is set to '8' i.e. if energy count crosses 8 digits, then it will reset to zero.

(2) Energy on Display is Auto-Ranging for details (see section 4.1)

3.2.1.13 Energy Rate

This screen allows user to enter energy update rate in minutes. After entering particular value in minutes, the energy will be updated on modbus location from 30145 to 30165 of 3X register and 40145 to 40165 of 4X register as per value that user has entered.



The user can select any integral value between 1 and 60 minutes.

Pressing the "▲" key accepts the present value and advances to "Impulse rate" screen (see Section 3.2.1.14).

Similarly, pressing the " \checkmark " key accepts the present value and advances to "Energy Digit Reset Count" screen (see Section 3.2.1.12).
Pressing the " ■ " key advances to the Energy Rate edit mode. Pressing " ▲ " and " ▼ " scrolls the count in minutes from 1 to 60 and from 60 to 1, respectively. Ex. If Energy Rate is set to 2 then energy will get stored after 2 minutes on the modbus.

Pressing " - " key sets the value selected and advances to the "Energy Rate" menu (see Section 3.2.1.13).

NOTE: Default value is set to '15' i.e. 15 min.

3.2.1.14 Impulse Rate



This screen shows user to Impulse rate depending upon the PT secondary and CT Secondary set by user. 'k' indicates value is multiplied by 1000.

Pressing the " ▲ " key advances to "quit" screen (see Section 3.2.1.1). Similarly, pressing the " ▼ " key advances to "Energy Rate" screen (see Section 3.2.1.13).

System nominal power	Impulse constant
<=400	16000
<=800	8000
<=1600	4000
<=3200	2000
>3200	1000

3.2.1.15 Quit System Parameters



This screen allows user to Exit from System Parameter selection setup.

Pressing the " A " key advances to "System Type" screen (see Section 3.2.1.1). Similarly, pressing the " v " key advances to "Impulse Rate" screen (see Section 3.2.1.14). Pressing the "-" key advances to "System Parameter Selection" screen (see Section 3.2.1).

3.2.2 Communication Parameter Selection

3.2.2.1 Address Setting



This screen applies to the RS 485 output only. This screen allows the user to set RS 485 address for the meter.

The allowable range of addresses is 1 to 247.

Press " A " key to advance to "RS 485 Baud Rate" screen (see Section 3.2.2.2) or press the " v " key to advance to the "Quit Communication Parameters" screen (see Section 3.2.2.4).



Press "- " to enter into edit mode, prompt for first digit.

Press the "A " and " V " keys to scroll the value of the first digit. Press the "-" key to advance to next digit.

Similarly, enter second and third digits of address. After entering third digit, pressing " - " key confirms the selection and shows "Address Setting" screen (see Section 3.2.2.1).

The default setting is '1'.

3.2.2.2 RS 485 Baud Rate



This screen allows the user to set Baud Rate of RS 485 port. The values displayed on screen are in kbaud.

Pressing " \blacktriangle " key accepts the present value and advance to the "RS 485 Parity Selection" screen (see Section 3.2.2.3) and pressing the " \blacktriangledown " key accepts the present value and advance to the "Address Setting" screen (see Section 3.2.2.1).

Pressing the "— " key advances to the "Baud Rate Edit" mode and "▲" & "▼" keys scrolls the value through 4.8, 9.6, 19.2, 38.4 and 57.6 kbaud.

Pressing the "....." key sets the value and shows the "RS 485 Baud Rate" screen (see Section 3.2.2.2).

3.2.2.3 RS 485 Parity

This screen allows the user to set Parity & number of stop bits of RS 485 port.



Pressing "▲" key accepts the present value and advances to "Quit Communication Parameters" screen (see section 3.2.2.4).Similarly, pressing "▼" key accepts the present value and advances to "RS 485 Baud Rate" screen (see section 3.2.2.2).

Pressing the " — " key advances to the "Parity & Stop bit Edit" mode & keys "▲ " and " ▼ " scrolls the value through: no 1 : no parity with one stop bit is : even parity with one stop bit odd : odd parity with one stop bit

Pressing " — " key sets the value and advances to "RS 485 Parity Selection" screen (see Section 3.2.2.3). NOTE: Default value is set as 'no 1'.

3.2.2.4 Quit Communication Parameters



This screen allows user to exit from system "Communication Parameter Selection" setup.

Pressing the "▲" key advances to "Communication Parameter Selection" screen (see Section 3.2.2.1). Similarly, pressing the "▼" key advances to "RS 485 Parity" screen (see Section 3.2.2.3).

Pressing the " — " key advances to "Communication Parameter Selection " screen (see Section 3.2.2).

3.2.3 IP Parameter Selection

3.2.3.1 IP Address Setting



This screen allows the user to set Ip address.

Pressing " \blacktriangle " key accepts the present value and advance to the "Quit Ip parameters" screen (see Section 3.2.3.5) and pressing the " \checkmark " key accepts the present value and advance to the "Subnet Mask Setting" screen (see Section 3.2.3.2).

Pressing the "- " key advances to the "IP address Edit" mode

Pressing the "- " key sets the value and shows the "IP address Edit Screen" (see Section 3.2.3.1).



Pressing the "▲ " or " ▼ " key scrolls the value of the most significant digit from 0 through 9 or 9 through 0, respectively

Pressing the " = " key accepts the present value at the cursor position and advances the cursor to the next less significant digit.

user can set value from 000.000.000.000 to 255.255.255.255.

After entering the least significant digit, pressing the " - " key sets the value

3.2.3.2 Subnet Mask Setting



This screen allows the user to set Subnet Mask.

Pressing "▲ " key accepts the present value and advance to the "IP address setting" screen (see Section 3.2.3.1) and pressing the '▼ " key accepts the present value and advance to the "Default gateway Setting" screen (see Section 3.2.3.).

Pressing the "-" key advances to the "Subnet Mask Edit" mode



Pressing the "▲ " or " ▼ " key scrolls the value of the most significant digit from 0 through 9 or 9 through 0, respectively Pressing the " key accepts the present value at the cursor position and advances the cursor to the next less significant digit. user can set value from 000.000.000 to 255.255.255.255. After entering the least significant digit, pressing the " ● " key sets the value

3.2.3.3 Default Gateway Setting



This screen allows the user to set Subnet Mask.

Pressing "▲ " key accepts the present value and advance to the "Subnet Mask setting" screen (see Section 3.2.3.2) and pressing the "♥ " key accepts the present value and advance to the "Server port Setting" screen (see Section 3.2.3.4).

Pressing the "-" key advances to the "Default Gateway Edit" mode



Pressing the "▲ " or " ▼ " key scrolls the value of the most significant digit from 0 through 9 or 9 through 0, respectively

Pressing the " = " key accepts the present value at the cursor position and advances the cursor to the next less significant digit.

user can set value from 000.000.000.000 to 255.255.255.255.

After entering the least significant digit, pressing the " - " key sets the value

3.2.3.4 Server Port Settings



This screen allows the user to set Server Port for the meter. The allowable range of addresses is 001 to 999.

Press " ▲ " key to advance to "Default Gateway" screen (see Section 3.2.3.3) or press the " ▼ " key to advance to the "Quit IP Parameters" screen (see Section 3.2.3.5).



Press "- " to enter into edit mode, prompt for first digit.

Press the " **A** " and " **v** " keys to scroll the value of the first digit. Press the " **—** " key to advance to next digit.

Similarly, enter second and third digits of address. After entering third digit, pressing " —" key confirms the selection and shows "Server Port Setting" screen (see Section 3.2.3.4).

The default setting is '502'.

3.2.3.5 Quit IP Parameters



This screen allows user to exit from system " IP Communication Parameter Selection" setup.

Pressing the " ▼ " key advances to "Server Port Selection" screen (see Section 3.2.3.4). Similarly, pressing the " ▲ " key advances to "IP address Selection" screen (see Section 3.2.3.1).

Pressing the "- " key advances to "IP Parameter Selection " screen (see Section 3.2.3).

3.2.4 Reset Parameter Selection

3.2.4.1 Resetting Parameter

This screen allows the users to reset Energy, Lo(Min), hi(Max), Demand, Run hour, On hour, No. of Interrupts, Load Profile and Time Datalog.

Reset Parameters Reset (None) " key advances to "Reset Parameters" screen. Pressina " Pressing the " A " or " V " key advances to "Reset Parameter Selection" rSEŁ screen (see section 3.2.4). nonE Pressing the "-" key advances to edit mode. CESEF Edit mode Pressing "▲ " & " ▼ " keys scroll through the parameters given below: ALL : reset all resettable parameters d : reset all demand parameters rSEŁ E : reset all energies nonE Hi : reset maximum values of voltage & current Lo : reset minimum values of voltage & current ЕЧ יF hr : reset run hour & on hour intr : reset no. of auxiliary supply interruption count rESEE time : reset the time based datalog buffers to store no values LoAd : reset the load profile datalog buffers to store no values

Pressing the " - " key advances to "PIs Wait" screen and resets the parameter selected followed by "Reset Parameters" screen.

1 Caution : When Demand or Energy parameter are reset ,it will reset load profile log data.

3.2.5. Output Option Selection

This screen applies to the relay output option selection. Pressing " - " key advances to "Relay Selection" menu (see Section 3.2.5.1).

3.2.5.1 Relay Selection



Pressing *▲ * and *▼ * keys scrolls through the following screens: rEL1 : To select options for relay 1 (See section 3.2.5.1.1). rEL2 : To select options for relay 2 (See section 3.2.5.1.1). guit : To exit the Output Options menu and give the *Output Option

quit : To exit the Output Options menu and give the "Output Option Selection" screen (see Section 3.2.5).

Pressing " — " key advances to Relay1 or 2 Output Selection menu (see Section 3.2.5.1.1).

3.2.5.1.1 Relay 1 or 2 Output Selection Menu



Pressing " - " key makes the following options available for relay1 and relay2:

1. Pulse : Relay in Pulse output mode (see Section 3.2.5.1.1.1)

2. Limit : Relay in Limit output mode (see Section 3.2.5.1.1.2)

3. Timer : Relay in Timer output mode (see Section 3.2.5.1.1.3)

4. RTC Relay : Relay in RTC output mode (see Section 3.2.5.1.1.4)

Press " \blacktriangle " and " \blacktriangledown " keys to navigate between the above options and press " \blacksquare " key to confirm the selection. The default option is set as 'Pulse'.

3.2.5.1.1.1 Pulse Output



This screen is used to set the pulse output parameter.

Pressing * ▲ * key advances to *Parameter Selection* screen (see Section 3.2.5.1.1.1) whereas pressing * ▼ * key advances to *Quit Pulse Output* menu (see Section 3.2.5.1.1.1.4).

3.2.5.1.1.2 Limit Output



This screen is used to assign Relay in Limit output mode.

Pressing "▲ " key shows "Parameter Selection" screen (see Section 3.2.5.1.2.1) whereas pressing "▼ " key shows the "Quit Limit Output" screen (see Section 3.2.5.1.1.2.7).

3.2.5.1.1.3 Timer



This screen is used to assign Relay in Timer output mode.

Pressing "▲" key will give the Number of Cycles menu (see Section 3.2.5.1.1.3.1) whereas pressing "▼" key gives the Quit Timer output menu (see Section 3.2.5.1.1.3.6).

3.2.5.1.1.4 RTC Relay



This screen is used to assign Relay in RTC output mode.

Pressing "▲" key advances to "Weekdays Selection" menu (see Section 3.2.5.1.1.4.1) whereas pressing "▼" key advances to "Quit RTC Output" menu (see Section 3.2.5.1.1.4.5).

3.2.5.1.1.1.1 Parameter Selection



This screen allows the user to assign energy for pulse output. Pressing " A " key accepts the present setting and advance to "Pulse duration selection" (see section 3.2.5.1.1.1.2) and pressing " ▼ " key accepts the present setting and advance to "Quit Relay Output" selection (see section 3.2.5.1.1.1.4).

Pressing the " - " key advances to edit mode and " A " and " V " keys scrolls through the energy setting:

Edit mode



- I E : Import Energy (Active)
- E E : Export Energy (Active)
- C rE : Capacitive Reactive Energy
- L rE : Inductive Reactive Energy
- A E : Apparent Energy

Pressing the " - " key sets the value & gives the "Parameter Selection" menu (see section 3.2.5.1.1.1.1).

NOTE: Default configuration is set as 'I-E'.

3.2.5.1.1.1.2 Pulse Duration



This screen applies only to the Pulse output mode of relay. This screen allows the user to set Relay energization time in milliseconds.

Pressing "A " key accepts the present value and advance to "Pulse Rate" screen (see section 3.2.5.1.1.1.3). Similarly, pressing "▼ " key accepts the present value and advance to "Parameter Selection" screen (see section 3.2.5.1.1.1.1).

Pressing the " - " key advances to "Pulse Duration Edit" mode and " - " and "v" keys scroll the value through 60, 100 and 200 ms.

Pressing the "- " key selects the value and advances to "Pulse Duration " menu (see Section 3.2.5.1.1.1.2). NOTE: Default value is set to '100' ms

3.2.5.1.1.1.3 Pulse Rate



This screen applies to the Pulse Output option only. The screen allows user to set the Energy Pulse Rate divisor. Divisor values can be selected through 1,10,100,1000 as per EnoP set. Refer TABLE 4 for details.

Pressing "▲ " key accepts the present selection and takes to the "Quit Pulse Output" menu (See section 3.2.5.1.1.1.4) and pressing "▼ " key accepts the present selection and takes to the "Pulse Duration" screen (see Section 3.2.5.1.1.1.2).



Pressing the "— " key advances to "Pulse Rate Divisor Edit" mode & keys "▲" and "▼" scrolls the value through the values 1,10,100 and 1000.

Pressing the " \blacksquare " key gives the "Pulse Rate" screen (see Section 3.2.5.1.1.1.3).

The default setting is '1'.

Note: If energy resolution is in kWh or Mwh then value is default 1.

3.2.5.1.1.1.4 Quit Pulse Output



The screen allows user to exit the Pulse Output selection menu.

Pressing " \blacktriangle " key advances to the "Pulse Output" menu (see Section 3.2.5.1.1.1) and pressing " \checkmark " key advances to the "Pulse Rate" menu (See section 3.2.5.1.1.1.3).

Pressing " — " key advances to the "Relay Selection" menu (see Section 3.2.5.1.).

3.2.5.1.1.2.1 Limit Output Parameters



This screen is for Limit output mode selection. It allows the user to set Limit output corresponding measured value. Refer TABLE 3 "Parameter for Limit output" for assignment.

Pressing "▲ " key accepts the present parameter and for 37/38/39/40/41 as present value, advances to the "Energy Count Configuration" screen (see section 3.2.5.1.1.2.2) whereas for other values, advances to the "Parameter Configuration" screen (see Section 3.2.5.1.1.2.5).

Whereas pressing " • " key accepts the present parameter and advances to the "Quit Limit Output" screen (see section 3.2.5.1.1.2.10).

Pressing the "- " key advances to "Limit Output Parameters" screen (see Section 3.2.5.1.1.2.1).

3.2.5.1.1.2.2 Energy Count Configuration



This screen is used to set the Limit Configuration for Energy Count.

Selecting 37/ 38/ 39/ 40/ 41 as Limit Output Parameter (see Section 3.2.5.1.1.2.1) allows the user select one of the following configurations: En (To Energize the Relay)

d-En (To De-Energized the Relay)

Pressing the " \blacktriangle " key accepts the present selection and advances to the "Energy Trip Point" screen (see section 3.2.5.1.1.2.3) and pressing the " \checkmark " key accepts the present selection and advances to the "Limit Output Parameters" screen (see section 3.2.5.1.1.2.1).

Pressing the "-" key advances to Energy Count Configuration edit mode and " \bullet " and " \bullet " keys scrolls through the modes available.

Pressing the " - " key sets the displayed value and advances to "Energy Count Configuration" Screen (see Section 3.2.5.1.1.2.2).

NOTE: Default configuration is set to 'En'.

3.2.5.1.1.2.3 Energy Trip Point



This screen is used to trip the relay using the energy count. The relay trips after the lapse of "ON Delay" time (see Section 3.2.5.1.1.2.4) from the moment the energy count reaches the value of Energy Trip Point set by the user in addition to its value at the moment the Energy Trip Point set.

Pressing the " \blacktriangle " key accepts the present value and advances to the "Energy Count ON Delay" screen (see Section 3.2.5.1.1.2.4) and pressing the " \checkmark " key accepts the present value and advances to the "Energy Count Configuration" screen (see Section 3.2.5.1.1.2.2).

Pressing the "- " key advances to Energy Count Configuration edit mode.

Energy Count Configuration edit mode



Press "- " key, prompt for the first digit.

Press the "▲" and "▼" keys to scroll the value between 0 and 9,whereas Press the "■" key to lock the present selection and advance to next digit.

Similarly, lock the value of all the remaining digits of the 7 digit count in a similar way until the last digit is reached.



Pressing the "_ " key for the last digit sets the value for Energy Trip Point.

For example, if the value set for Energy Trip Point is 888 and the value of the corresponding parameter at the moment this value is set is 1077, then the relay will trip after x sec of the moment the value of the parameter becomes 1965 (= 1077 + 888), where x is the ON Delay (see Section 3.2.5.1.1.2.4).

The value of Energy Trip Point can range from 10 to 99999999.

Default value is set to '10'.

Caution:Once the relay has tripped, then to reactivate the Energy Tripping function, the user has to either reset the energy or re-enter the energy count.

3.2.5.1.1.2.4 Energy Count ON Delay



This screen allows the user to set ON Delay time in **seconds** for Relay Limit Assigned Parameter. Refer Section 3.2.5.1.1.2.3 for details.

Pressing "▲ " key accepts the present value and advance to "Quit Limit Output" screen (see Section 3.2.5.1.1.2.10) and pressing "▼ " key accepts the present value and advances to "Energy Trip Point" screen (see Section 3.2.5.1.1.2.3).



Press "■" key, prompt for the first digit. Press the "▲" and "▼" keys to scroll the values of the first digit. Press the "■" key to advance to next digit.

Similarly, enter second, third and fourth digits also.

After the fourth digit is entered, pressing " — " key sets the value and advances to "Energy Count ON Delay" screen (see Section 3.2.5.1.1.2.4).

The value for this parameter can range from 0001 to 9999 seconds. Note: Default value is set to '1' second.

3.2.5.1.1.2.5 Parameter Configuration



Selecting Limit Output Parameter (see Section 3.2.5.1.1.2.1) other than 37/38/39/40/41 allows the user select one of the following configurations:

- Hi-E (High Alarm & Energized Relay)
- Hi-d (High Alarm & De-Energized Relay)
- Lo E (Low Alarm & Energized Relay)
- Lo d (Low Alarm & De-Energized Relay)



(For details refer to section 5.2)

Pressing the " \blacktriangle " key accepts the present selection and advances to the "Trip Point" screen (see section 3.2.5.1.1.2.6) and pressing the " \checkmark " key accepts the present selection and advances to the "Limit Output Parameters" screen (see section 3.2.5.1.1.2.1).

Pressing the "— " key advances to Parameter Configuration edit mode and "▲" and "▼" keys scrolls through the modes available.

Pressing the " — " key sets the selected config. and advances to "Limit Configuration" Screen (See section 3.2.5.1.1.2.2). Note: Default configuration is set to 'Hi-E'.

3.2.5.1.1.2.6 Trip Point

This screen applies to the Trip point selection for parameters other than 37/38/39/40/41 selected in Section 3.2.5.1.1.2.1.



This screen allows the user to set Trip point for instruments.

The allowable range is 10% to 120% for High Alarm, 10% to 100% for Low Alarm (refer TABLE 3).

Pressing the " \blacktriangle " key accepts the present value and advances to the "Hysteresis" screen (see section 3.2.5.1.1.2.7) and pressing the " \checkmark " key accepts the present value and advances to the "Parameter Configuration" screen (see Section 3.2.5.1.1.2.5).

Press "- " to confirm and go to "Trip Point" screen (see Section 3.2.5.1.1.2.6).

Pressing "■" key prompts for first digit. Press the "▲" and "▼" keys to scroll the values of the first digit. Press the "■" key to advance to next digit. Similarly, enter second and third digits also.

Note: Default value is set to '10' %.

3.2.5.1.1.2.7 Hysteresis



This screen applies to the Hysteresis selection. This screen allows the user to set Hysteresis for relay output.

The allowable range is 0.5% to 50.0 % of Trip point.

Pressing the " \blacktriangle " key accepts the present value and advances to the "Energizing Delay" screen (see Section 3.2.5.1.1.2.6) and pressing the" \checkmark " key accepts the present value and advances to the "Trip Point" screen (see Section 3.2.5.1.1.2.6).



Pressing " → " key prompts for first digit. Press the " ▲" and " ▼" key to scroll the value of the first digit. Press the " →" key to advance to next digit.

The second digit prompts. Press " ▲ " and " ▼ " keys to scroll through 0 and 9 and " ■ " key to set the digit and advance to the third digit. The third digit prompts. Press " ▲ " and " ▼ " keys to scroll through 0 and 9.

Press " _ " to confirm the value and advance to "Hysteresis" screen (see Section 3.2.5.1.1.2.7). Refer Section 5.2 for further details.

Note: Default value is set to '0.5' %.

3.2.5.1.1.2.8 Energizing Delay

This screen allows the user to set Energizing Delay time in seconds for Relay Limit Assigned Parameters.



Pressing " \blacktriangle " key accepts the present value and advances to "De-Energizing Delay" screen (see Section 3.2.5.1.1.2.9) and pressing " \checkmark " key accepts the present value and advances to "Hysteresis" screen (see Section 3.2.5.1.1.2.7).

Pressing the "- " key advances to "Energizing Delay" Edit mode.

Pressing " — " key sets displayed value & advances to Assignment of "Energizing Delay" menu (See section 3.2.5.1.1.2.8).



Pressing "- " key prompts for first digit..

Press the " A " and " V " key to scroll the value of the first digit. Press the " " " key to advance to second digit.

The second digit prompts. Press " \blacktriangle " and " \checkmark " keys to scroll through 0 and 9 and " \multimap " key to set the digit and advance to the third digit. The third digit prompts. Press " \blacktriangle " and " \checkmark " keys to scroll through 0 and 9 and " \multimap " key to set the digit and advance to the fourth digit. Press \clubsuit " and " \And " keys to scroll through 0 and 9. Press " \multimap " to confirm the value and advance to "Energizing Delay" screen (see Section 3.2.5.1.1.2.8).

The value of Energizing Delay can be set between 1 and 9999 seconds. Note: Default value is set to '1' second.

3.2.5.1.1.2.9 De-Energizing Delay

This screen allows the user to set De-Energizing Delay time in seconds for Relay Limit Assigned Parameters.

Pressing " A " key accepts the present value takes to "Quit Limit Output" dEn menu (See section 3.2.5.1.1.1.2.10). Similarly, pressing " v " key accepts the present value takes to Energizing del y Delay" menu (See section 3.2.5.1.1.1.2.8). Pressing the "- " key advances to "De-Energizing Delay" Edit mode. Pressing "- " key sets displayed value and takes back to "De-Energizing Delay" screen (see Section 3.2.5.1.1.2.9). Pressing "- " key prompts for first digit. Press the "A " and " V " key to scroll the value of the first digit. En Press the "- " key to advance to second digit. dEL Y The second digit prompts. Press "▲ " and " ▼ " keys to scroll through 0 00 IO and 9 and "- " key to set the digit and advance to the third digit. The third digit prompts. Press "▲ " and "▼ " keys to scroll through0 and 9 ደብ יF and "- " key to set the digit and advance to the fourth digit. out Press "▲" and " ▼ " keys to scroll through 0 and 9. Press "- " key to confirm the value and advance to "De-Energizing Delay" screen (see Section 3.2.5.1.1.2.9).

The value of De-Energizing Delay can be set between 1 and 9999 seconds. Note: Default value is set to '1' second.

3.2.5.1.1.2.10 Quit Limit Output



The screen allows user to exit the Relay output selection menu.

Pressing "▲" key advances to "Limit Output" menu (See section 3.2.5.1.1.2) and pressing "▼" key advances to "De-Energizing Delay" menu (see Section 3.2.5.1.1.2.9) if Limit Output Parameter (see Section 3.2.5.1.1.2.1) set is not 37/38/39/40/41, otherwise it advances to "Energy Count ON Delay" screen (see Section 3.2.5.1.2.4).

Pressing " \blacksquare " key advances to "Relay Selection" menu (see Section 3.2.5.1).

3.2.5.1.1.3.1 Number of Cycles



The value decides how many times the timer will repeat the switching after it has been started in the **timer based relay output option**.

Pressing "▲" key confirms the value and advances to the "Timer Configuration" menu (See section 3.2.5.1.1.3.2) and pressing "▼" key advances to "Timer" menu (See section 3.2.5.1.1.3).

The value for this **parameter can range from 0000 to 9999.** If the value is set as 0000, the timer will keep repeating the cycles until 9999 cycles are complete or the timer is stopped by the user. Refer Section 5.3 for more details.



Press "- " key, prompt for the first digit.

Press the "▲ " and "▼ " keys to scroll the values of the first digit. Press the "— " key to advance to next digit. Similarly, enter second, third and fourth digits also.

After the fourth digit has been entered, pressing "—" key sets the value and advances to "Number of Cycles" screen (see Section 3.2.5.1.1.3.1). The default setting is '10' cycles.

3.2.5.1.1.3.2 Timer Configuration



The option decides the relay configuration for timer output. Two options are available:

1. En : Energize on start

2. d-En : De-energize on start.

Pressing "▲" key confirms the selection and advances to the "On Delay" menu (See section 3.2.5.1.1.3.3) and pressing "▼" key advances to the "Number of Cycles" menu (See section 3.2.5.1.1.3.1).



Press " \blacksquare " key to enter the edit mode and press " \blacktriangle " and " \blacktriangledown " keys to navigate between the options.

Pressing " — " key sets the selected config. and advances to the Timer Configuration menu (see Section 3.2.5.1.1.3.2).

The default setting is 'En'.

3.2.5.1.1.3.3 On Delay



The value decides the time in seconds taken by the relay in timer configuration before tripping after it is started.

Pressing "▲ " key confirms the value and advances to the "Off Delay" menu (See section 3.2.5.1.1.3.4) and pressing "▼" key advances to the "Timer Configuration" menu (See section 3.2.5.1.1.2).

Press "- " key, prompt for the first digit.

Press the "A " and " V " keys to scroll the values of the first digit.



Press the "—" key to advance to next digit. Similarly, enter second, third and fourth digits also. After the fourth digit is entered, pressing "—" key sets the value and advances to "On Delay" screen (see Section 3.2.5.1.1.3.3).

The value for this parameter can range from 0001 to 9999 seconds. The default value is '10' seconds.

3.2.5.1.1.3.4 Off Delay



The value decides the time in seconds taken by the relay in timer configuration before coming out of the trip state after it has tripped.

Pressing " \blacktriangle " key confirms the value and advances to the "Quit Timer Output" menu (See section 3.2.5.1.1.3.5) and pressing " \checkmark " key advances to the "On Delay" menu (See section 3.2.5.1.1.3.3).



Press the "- " key to advance to next digit.

Similarly, enter second, third and fourth digits also. After the fourth digit has been entered, pressing " — " key sets the value and advances to Off Delay screen (see Section 3.2.5.1.1.3.4).

The value for this parameter can range from 0001 to 9999 seconds.

The default value is '10' seconds.

3.2.5.1.1.3.5 Quit Timer Output



The screen allows user to exit the Timer output menu.

Pressing " A " key advances to the "Timer Output" menu (see Section 3.2.5.1.1.3) and pressing " ▼ " key advances to "Off Delav" menu (see Section 3.2.5.1.1.3.4).

Pressing " - " key advances to "Relay Selection" menu (see Section 3.2.5.1).

3.2.5.1.1.4 RTC Relav

3.2.5.1.1.4.1 Weekdays Selection



This screen allows user to select the days of the week on which the relay behaves as configured for RTC Relay settings.

Pressing " A " key confirms the selection and advances to the "Relay Configuration" menu (See section 3.2.5.1.1.4.2) and pressing " ▼ " key advances to the "RTC Relay" menu (See section 3.2.5.1.1.4).

Pressing "- " key advances user to the "Edit Weekdays" mode where the user can edit the working weekdays selection.

Edit Weekdavs



Press " _ " key, prompt for the first digit.

The first digit at the lower row of the screen represents SUNDAY. Press the "A " and " V " keys to scroll the value between 0 and 1, where 0 : Relay is not activated for the weekday selected

- 1 : Relay is active for the weekday selected.

Press the "- " key to lock the present selection and advance to next digit representing MONDAY.



Lock the selection for all the remaining days, till selection for the last day, i.e., SATURDAY, is reached.

Once the selection for SATURDAY is set by pressing " — " key, the "Weekdays Selection" screen appears again (see Section 3.2.5.1.1.4.1) and sets the days for relay to be active/deactive.

The default setting is '1111111' i.e., active for all the days.

3.2.5.1.1.4.2 Relay Configuration



The option decides the relay configuration in timer mode. Two options are available:

1. En : Energize on start

2. d-En : De-energize on start.

Pressing " \blacktriangle " key confirms the selection and advances to the "On Time" menu (See section 3.2.5.1.1.4.3) and pressing " \checkmark " key confirms the selection and advances to "Weekdays Selection" menu (see Section 3.2.5.1.1.4.1).

Press " \blacksquare " key to enter the edit mode and press " \blacktriangle " and " \bigtriangledown " keys to navigate between the options.

Pressing " — " key sets the selected option and advances to Relay Configuration menu (see Section 3.2.5.1.1.4.2).

The default setting is 'En', i.e., energized on start.

3.2.5.1.1.4.3 ON Time



Edit ON Time



On Time is the time at which the relay becomes active. The time is displayed in HH:MM format and its range is 00:00 to 23:59.

Pressing "▲" key confirms the value and advances to the "OFF Time" menu (See section 3.2.5.1.1.4.4) and pressing "▼" key advances to the "Relay Configuration" menu (See section 3.2.5.1.1.4.2).

Pressing "- " key advances to the Edit ON Time option.

Press "= ", prompt for 10's place of HH. Press "= ", prompt for 1's place of HH. Press "= ", prompt for 10's place of MM. Press "= ", prompt for 1's place of MM.

Keys "▲ " and "▼ " are used to change the corresponding values.

Pressing "—" key confirms the selection and advances to the "ON Time" menu (see Section 3.2.5.1.1.4.3). The default setting is '06:00', i.e., 6 A.M.

3.2.5.1.1.4.4 OFF Time



OFF Time is the time at which the relay deactives. The time is displayed in HH:MM format and its range is 00:00 to 23:59.

Pressing "▲ " key confirms the value and advances to the "Quit RTC Output" menu (See section 3.2.5.1.1.4.5) and pressing "▼" key advances to the "ON Time" menu (See section 3.2.5.1.1.4.3).

Pressing "- " key advances to the "Edit OFF Time" option.

Edit OFF Time



Press " , prompt for 10's place of HH. Press "- ", prompt for 1's place of HH.

Press "- ", prompt for 10's place of MM.

Press "-", prompt for 1's place of MM.

Keys "▲" and "▼" are used to change the corresponding values.

Pressing "-" key confirms the selection and advances to the "OFF Time" menu (see Section 3.2.5.1.1.4.4).

The default setting is '18:00' i.e., 6 P.M.

3.2.4.1.1.4.5 Quit RTC Output



The screen allows user to exit the RTC output menu.

Pressing " A " key advances to the "RTC Output" menu (see Section 3.2.5.1.1.4) and pressing " ▼ " key advances to the "OFF Time" menu (see Section 3.2.5.1.1.4.4).

Pressing " - " key advances to the "Relay Selection" (see Section 3.2.5.1).

3.2.6 Datalog Option Selection

3.2.6.1 Event Based Datalog Setup



This screen is used to enter into event datalogging feature.

Pressing the " A " key advances to the "Time Based Datalog" menu (see Section 3.2.6.2) and pressing the " v " key takes to the "Quit Datalog Option" menu (see Section 3.2.6.4).

Pressing the "- " key advances to the Event Based datalog selection and pressing the "A " and "V " key scrolls through the options available: datalog : YES/no (see Section 3.2.6.1.1) quit : to exit Event Datalog Selection (see Section 3.2.6.1.2)

3.2.6.1.1 Event Based Datalog Selection



This screen is used to start or stop event datalogging.

Pressing the " - " key allows the user to start or stop event based datalogging by selecting "YES" or "no", respectively using " \blacktriangle " and " \blacktriangledown " keys.

Once the required option is selected, pressing the " — " key sets the selection and advances to the Event Based Datalog selection screen (see Section 3.2.6.1.1).

3.2.6.1.2 Quit Event Datalog



This screen is used to exit event based datalog selection.

Pressing the " — " key advances to the Event Based Datalog setup screen (see Section 3.2.6.1).

3.2.6.2 Time Based Datalog Setup



This screen is used to enter into time based datalog feature.

Pressing the "▲" key takes to the "Load Profile Datalog" menu (see section 3.2.6.3) and pressing the "▼" key takes to the "Event Based Datalog" menu (see section 3.2.6.1).

Pressing the " → " key advances to Time Based datalog selection and pressing " ▲ " and " ▼ " keys scrolls through various parameters related to time based datalogging (see Section 3.2.6.2.1 to Section 3.2.6.2.5).

3.2.6.2.1 Time Based Datalog Selection



This screen is used to start or stop time based datalogging.

Pressing the " \blacktriangle " key confirms the selection and advances to "Time Interval Selection" screen (see Section 3.2.6.2.2) and pressing the " \checkmark " key confirms the selection and advances to "Quit Time Based Logging" screen (see Section 3.2.6.2.5).

Pressing the " — " key allows the user to start or stop time based datalogging by selecting "YES" or "no", respectively using " \blacktriangle " and " \blacktriangledown " keys.

Pressing the " — " key sets the selection and if the selection is "YES", then "PIs Wait" screen appears followed by "Time Based Datalog Selection" screen (see Section 3.2.6.2.1).

Caution: The settings for time based logging (see Section 3.2.6.2.2 - Section 3.2.6.2.4) are not editable if time based datalog selection is set to YES (see Section 3.2.6.2.1).

3.2.6.2.2 Time Interval Selection



This screen is used to decide the time interval between two successive time datalog entries.

The allowable range is 01 - 60 minutes.

Pressing the *▲ * key confirms the selection and advances to *Parameter Count' screen (see Section 3.2.6.2.3) and pressing the *▼ * key confirms the selection and advances to *Time Based Datalog Selection* screen (see Section 3.2.6.2.1).



Press "—" to enter different time interval, it prompts for first digit. Press the "▲" and "▼" keys to scroll the value of the first digit. Press the "—" key to advance to next digit.

Similarly, enter the second digit of interval.

After entering second digit, pressing " — " key sets the value and advances to the "Time Interval Selection" screen (see Section 3.2.6.2.2).

The default value is '1' second.

3.2.6.2.3 Parameter Count



PRrR

Ent

Edit

dRERLoG

This screen is used to decide the number of parameters that will be logged in time based datalogging. The allowable range is 01 - 30.

Pressing the " A " key confirms the selection and advances to "Parameter Selection" screen (see Section 3.2.6.2.4) and pressing the " ▼ " key confirms the selection and advances to "Time Interval Selection" screen (see Section 3.2.6.2.2).

Press "- " to enter the parameter count, prompt for first digit.

Press the "A " and " V " keys to scroll the value of the first digit. Press the "- " key to advance to next digit.

Similarly, enter the second digit of interval. After entering second digit, pressing " - " key sets the value and advances to the "Parameter Count" screen (see Section 3.2.6.2.3). The default value is '1'

3.2.6.2.4 Parameter Selection

06



This screen is used to select the measurement parameters to be recorded. The allowable values are shown in Table 5.

For each of the parameter count set in Section 3.2.6.2.3, the corresponding parameter number (Refer Table 5) can be set by the user.

Pressing "
" key confirms the selection and allows the user to proceed for setting the next parameter until the last parameter is set which is followed by the "Quit Time Based Datalog" screen (see Section 3.2.6.2.5).

Where as pressing the "v " key confirms the selection and takes to the previous parameter set until the first parameter is reached which is followed by the "Parameter Count" screen (see Section 3.2.6.2.3).





Press "-" to enter the parameter selection.

Press the "▲" and "▼" keys to scroll the value of the measurement parameter number in decreasing and increasing order, respectively. Pressing "■" key sets the value and take user to the "Parameter Selection" screen (see Section 3.2.6.2.4) for the parameter set.

The default value is '000', i.e. no parameter to be logged.

3.2.6.2.5 Quit Time Based Datalog



This screen is used to exit time based datalog selection.

Pressing the "-" key advances to the "Time Based Datalog Setup" screen (see Section 3.2.6.2).

3.2.6.3 Load Profile Datalog Setup



This screen is used to enter into Load Profile datalog feature.

Pressing the "▲" key takes to the "Quit Datalog Option" menu (see Section 3.2.6.4) and pressing the "▼" key advances to "Time Based Datalog Setup" (see Section 3.2.6.2).

Pressing the " — " key advances to the Load Profile datalog selection and pressing the " ▲ " and " ▼ " key scrolls through the options available: datalog : YES/no (see Section 3.2.6.3.1) quit : to exit Load Profile Datalog selection (see Section 3.2.6.3.2)

3.2.6.3.1 Load Profile Datalog Selection



This screen is used to start or stop Load Profile datalogging.

Pressing the " \blacksquare " key allows the user to start or stop Load Profile datalogging by selecting "YES" or "no", respectively using " \blacktriangle " and " \blacktriangledown " keys.

Pressing the " - " key sets the selection and if the selection is "YES", then "PIs Wait" screen appears followed by "Load Profile Datalog Selection" screen (see Section 3.2.6.3.1). If the selection is "no", then "PIs Wait" screen does not appear.

1 Caution: If any setting is done by user which resets energy or demand parameter it will reset load profile data.

3.2.6.3.2 Quit Load Profile Datalog Selection



This screen is used to exit Load Profile datalog selection.

Pressing the "-" key advances to the "Load Profile Datalog Setup" screen (see Section 3.2.6.3).

3.2.6.4 Quit Datalog Option



The screen allows user to exit the Datalog Option menu.

Pressing "▲ " key advances to the "Event Based Datalog" menu (see Section 3.2.6.1) and pressing "▼" key advances to the "Load Profile Datalog Setup" menu (see Section 3.2.6.3).

Pressing " - " key advances to the "Datalog Option Selection Menu"(see Section 3.2.6).

3.2.7 Display Parameters

3.2.7.1 Feature Selection Menu



This menu allows the user to scroll through different Display Configurable features: bCl t : backlit on/off

Scrn : user screen on/off

Scrn : user screen on/off

Cont : Contrast level

quit : to exit Display Parameters screen

Pressing the " $_$ " key advances to the listed features (see Section 3.2.7.1.1 to Section 3.2.7.1.3).

3.2.7.1.1 Backlit



This screen allows the user to switch the backlit on or off.

Pressing the "▲ " and " ▼ " keys advances to "User Assignable Screens" (see Section 3.2.7.1.2) and "Quit Display Parameters" menu (see Section 3.2.7.1.4), respectively.

Pressing the " — " key shows the present status as on/OFF and pressing " — " key allows editing it whereas " ▲ " and " ▼ " keys advance to the "Backlit" menu (see Section 3.2.7.1.1).



In Edit Mode, pressing "▲" and "▼" keys allows the user to scroll between On/OFF and pressing "—" key confirms the selection.

Note: When backlit is switched 'Off', on pressing any key backlit will turn 'On' for 1 min.

Default value is set to 'On'.

3.2.7.1.2 User Assignable Screens



This screen allows the user to turn enable or disable the User Screen feature.

Using this feature, the user can select upto 10 measurement screens of his choice and scroll through only those selected screens.

Pressing the " ■ " key allows the user to advance to the "Screen Number Selection" menu (see Section 3.2.7.1.2.1) whereas pressing the " ▲ " and " ▼ " keys advances to "Backlit" menu (see Section 3.2.7.1.3) and "Contrast" menu (see Section 3.2.7.1.1) respectively.

3.2.7.1.2.1 Screen Number Selection



Pressing the * ▲ * key sets the present value for the number of screens to be shown and advance towards the "User Screens Selection" menu (see Section 3.2.7.1.2.2) whereas pressing the * ▼ * key sets the present value and advance to "Quit Userscreens" menu (see Section 3.2.7.1.2.3).

Pressing " — " key allow the user to set a different value for the number of user assignable screens using "▲" and "▼" keys.

The user can set the number of screens from 1 to 10.



Pressing " _ " key sets the selected value and advances to "Screen Number Selection" screen (see Section 3.2.7.1.2.1).

Note: 1.The value 0 should be chosen if the user wants all the screens to be shown.

 $\mathbf{2}.$ If User Screen feature is ON and System type is changed, then the Userscreen is disabled.

The default setting is '0' ,i.e., all screens are shown.

3.2.7.1.2.2 User Screens Selection



Pressing the " \blacktriangle " key confirms the selection and allows the user to proceed for setting the next userscreen until the last userscreen is set which is followed by the "Quit UserScreens" menu (see Section 3.2.7.1.2.3)

Whereas pressing the " \checkmark " key confirms the selection and take to the previous userscreen set until the first userscreen is reached which is followed by the "Screen Number Selection" screen (see Section 3.2.7.1.2.1).



Pressing the " - " key advances the User Screen Edit mode and pressing " \land " and " \checkmark keys scroll the value as per TABLE 1 "Measurement Screens".

Pressing " — " key sets the displayed value & advance to User Screen Selection (see Section 3.2.7.1.2.2) for the corresponding screen number.

3.2.7.1.2.3 Quit Userscreens



This screen is used to exit User defined Screen selection.

Pressing the " - " key advances to the "User Assignable Screens" menu (see Section 3.2.7.1.2).

3.2.7.1.3 Contrast



This screen allows the user to set the contrast for the display.

Pressing the " \blacktriangle " and " \blacktriangledown " keys advances to"Quit Display" menu (see Section 3.2.7.1.4) and "User Assignable Screens" menu (see Section 3.2.7.1.2), respectively.

Pressing the "—" key shows the present contrast value and pressing "—" key again will allow editing it whereas "▲" and "▼" keys advances to the "Contrast" menu (see Section 3.2.7.1.3).



In Edit Mode, pressing "▲ " and "▼ " keys allows the user to scroll between **contrast levels ranging from 1 to 4** and pressing " — " key confirms the selection.

Pressing " — " key advances to editing mode whereas pressing " ▲ " or " ▼" keys advances to the "Contrast" menu (see Section 3.2.7.1.3). Default value is set to '3'.

3.2.7.1.4 Quit Display Parameters



This screen allows user to Exit from User Assignable Feature selection setup.

Pressing the "▲ " key advances to "Backlit" screen. (see Section 3.2.7.1.1). Whereas pressing the "▼" key advances to "Contrast" screen. (see Section 3.2.7.1.3).

Pressing the " - " key advances to "Display Parameters" (see Section 3.2.7).

3.2.8 RTC Setting Screen



This screen allows the user to set date and time for the device RTC.

Pressing " - " advances to Date Settings (see Section 3.2.8.1) and pressing "▲ " and " ▼ " keys advances to Time Settings (see Section 3.2.8.2) and "Quit RTC" screen (see Section 3.2.8.3), respectively.

3.2.8.1 Date Settings



This screen allows the user to set the date for device RTC. The date is displayed in DD-MM-YY format in the settings and its range is 01-01-00 to 31-12-99 (for the 21st century, i.e., YY = 00 represents

2000 and YY = 99 represents 2099).

Press "- ", prompt for DD.

Press "— ", prompt for MM. Press "— ", prompt for YY.

Kevs "▲ " and " ▼ " are used to change the values of DD. MM and YY. After YY is set, pressing " - " advances to "PIs Wait" screen followed by "Date Settings" screen (see Section 3.2.8.1).

Caution : If date setting is changed then load profile log data will reset

3.2.8.2 Time Settings



This screen allows the user to set the time for device RTC. The date is displayed in HH:MM:SS format in the settings and its range is

00:00:00 to 23:59:59.

- Press "- ", prompt for 10's place of HH.
- Press "- ", prompt for 1's place of HH.

Press "-", prompt for 10's place of MM. Press "-", prompt for 1's place of MM.

Press "-", prompt for 1's place of SS

Press "-", prompt for 1's place of SS

Keys "▲ " and " ▼ " are used to change the corresponding values.

After 1's place of SS is set, pressing "-" advances to "Time Settings" screen (see Section 3.2.8.2).

3.2.8.3 Quit RTC



This screen allows user to Exit from RTC settings.

Pressing the "▲" key advances to "Date Settings" screen. (see Section 3.2.8.1). Whereas pressing the "▼" key advances to "Time Settings" screen (see Section 3.2.8.2).

Pressing the " - " key advances to "RTC Setting screen" (see Section 3.2.8).

3.2.9 Factory Reset Screen



This screen allows the user to erase all data from the meter and set all setup parameters to their default values.

Pressing the " - " key advances to the "Sure" (confirmation) screen which displays a "no".

Pressing "▲" or "▼" key advances to Factory Reset Screen (see Section 3.2.9).

Whereas pressing " \blacksquare " key advances to the Factory Reset selection screen.

Factory Reset selection screen



Pressing "▲" or "▼" key allows the user to select between "YES" or "no".

YES : Allow Factory Reset no : Don't allow Factory Reset

Pressing "—" accepts the selection and if the selection is "YES", advances to "PIs Wait" screen followed by the "Sure" screen of "Factory Reset Screen" (see Section 3.2.9). If the selection is "no", then "PIs Wait" screen does not appear.

3.2.10 Quit



This screen allows the user to set the meter to exit the setup menu.

Pressing " - " key quits from the Setup menu and advance to measurement screen at which the setup screen was accessed.

4. Energy Auto-Ranging On Display :

4.1 Calculating Display energy (Auto-ranging):

In case of energy auto-ranging the energy shown on modbus will be in terms of Overflow count (OF) & main energy counter (M) based on energy resolution & energy digit reset count (EDRC). So energy on modbus will have unit set in energy resolution on modbus (ie. Wh or KWh or MWh) & overflow energy counter (OF) on modbus will increment when main energy counter (M) on modbus will cross the number of digits set in energy digit reset count (EDRC) register (ie 7 or 8 or 9).

Whereas energy shown on display will be Combined energy (ie. Calculated from overflow count (OF) on modbus & main energy Count (M) on modbus) autoranged based on the 9 digits available on display to show energy.

Hence the purpose of this document is to establish a relation between modbus energy counters & display energy.

Total Energy = OF x 10^(EDRC) + M

Where: OF => Overflow Count on modbus

Modbus registers for Overflow count (OF):

Wh import Overflow count	=> 30109 Or 40109
Wh export Overflow count	=> 30113 Or 40113
VARh Capacitive Overflow count	=> 30117 Or 40117
VARh Inductive Overflow count	=> 30121 Or 40121
Vah Overflow count	=> 30125 Or 40125
M => Main Energy Counter on modbus

Modbus registers for Main energy Count (M):

Wh import	=> 30111 Or 40111
Wh export	=> 30115 Or 40115
VARh Capacitive	=> 30119 Or 40119
VARh Inductive	=> 30123 Or 40123
Vah	=> 30127 Or 40127

EDRC => Energy digit reset count set by user from display or modbus For display setting refer Section 3.2.1.12. For modbus refer Register 46039

Example for Calculating Display Energy from modbus Energy Counters:

Energy resolution on modbus register 46005 is set to 2 (ie. It will measure energy in kWh).

Overflow count (OF) value on modbus register 30109 for watt Import is 25.

Energy main counter (M) value on modbus register 30111 for watt import is 2587413.189

Energy digit reset count (EDRC) set by user in modbus register 46039 is 8. Then,

Total Energy = 25 x 10⁽⁸⁾ + 2587413.189 = 2500000000 + 2587413.189 = 2502587413.189 kWh

Therefore Energy shown on display will be: = 2502587.41 MWh

As maximum number of digits for displaying energy present on display is 9 and the value of Total Energy in above example exceeds 9 counts so Energy in kWh is converted to MWh for displaying as shown above.

Note: As maximum energy shown on the display is XXXXXXXX (9 Count) MWh so if count of above calculation for Total energy exceeds XXXXXXXX (9 Count) MWh then the display will show "---oF---" with unit MWh that indicates display energy has overflown. This condition will only get reset if the Total Energy Count gets reset.

Overflow count (OF) on modbus will reset to 0 if it exceeds 99999999 in case of Wh resolution, 99999 in case of kWh resolution & 99 in case of Mwh resolution.

4.2 Entering Energy start count:

(Applicable for meter with modbus or Ethernet)

Energy start count is the feature that allows user to set count from which the meter should start incrementing energy. The start count can be set for main energy start counter & Overflow energy start Counter separately from modbus or Ethernet. Setting of energy start count is dependent on Energy resolution & Energy digit reset count (EDRC).

Energy resolution on modbus can be set by user from display (Refer page 29 of user manual) or from modbus (refer modbus register 46005).

Value for Energy resolution can be set to Wh,kWh or MWh.

Minimum resolution of energy that can be set in energy start count register is, the Energy resolution set by user.

For Eg.: Energy start count can be set to minimum 1Wh if the value on modbus register 46005 is set to Wh Or Energy start count can be set to minimum 1MWh if the value on modbus register 46005 is set to MWh.

Maximum value that can be set in the main energy start count register is (10[^](EDRC)-1).

Where:

EDRC => Energy digit reset count set by user from display or modbus For display setting refer Section 3.2.1.12. For modbus refer Register 46039

Example for calculating maximum limit of main energy start count :

If EDRC is 7 then Maximum value that can be set in the main energy start count (M) register is

= (10^(EDRC)-1) = ((10^7)-1) = (10000000-1) = 9999999

Similarly if EDRC is 8 the maximum value is 999999999 & if EDRC is 9 maximum value is 999999999.

Maximum value that can be set in the Overflow energy start count register is 99999999 for Wh renergy resolution ,99999 for kWh energy resolution ,99 for MWh resolution.

Note: Energy start Count can only be entered in integer value.

5. Relay Output (Optional) :

The Meter is provided with relay for pulse output for limit switch as a Timer as well as RTC Relay.

5.1 Pulse Output :

Pulse Output is the potential free, very fast acting relay contact which can be used to drive an external mechanical counter for energy measurement. The Pulse Output can be configured to any of the following parameter through setup parameter screen:

3) Capacitive Reactive Energy

- 1) Active Energy (Import)
- 2) Active Energy (Export)

- 4) Inductive Reactive Energy 5) Apparent Energy

TABLE 4 : Energy Pulse Rate Divisor

1.For Energy Output in Whr

2.For Energy Output in kWhr

	Pulse rate				
Divisor	Pulse	System Powe			
1	1 per kWhr	Up to 3600 k			
	1 per MWhr	Above 3600			

3.For Energy Output	in
MWhr	

	Pulse rate
Divisor	Pulse
1	1 per MWhr

	Pul	se rate			
Divisor	Pulse	System Power*			
	1 per Whr	Up to 3600 W			
1	1 per kWhr	Up to 3600 kW			
	1 per MWhr	Above 3600 kW up to 30000 kW			
	1 per 10Whr	Up to 3600 W			
10	1 per 10kWhr	Up to 3600 kW			
	1 per 10MWhr	Above 3600 kW up to 30000 kW			
100	1 per 100Whr	Up to 3600 W			
	1 per 100kWhr	Up to 3600 kW			
	1 per 100MWhr	Above 3600 kW up to 30000 kW			
1000	1 per 1000Whr	Up to 3600 W			
	1 per 1000kWhr	Up to 3600 kW			
	1 per 1000MWhr	Above 3600 kW up to 30000 kW			
Pulse	Pulse Duration 60 ms,100 ms or 200 ms				

Above options are also applicable for Apparent and Reactive Energy.

*Note:

1) System power = 3 x CT(Primary) x PT (Primary) L-N for 3 Phase 4 Wire

- 2) System power = Root3 x CT(Primary) x PT (Primary)L-L for 3 Phase 3 Wire
- 3) System power = CT(Primary) x PT(Primary)L-N for 1 Phase 2 Wire

5.2 Limit Switch :

Limit switch can be used to monitor the measured parameter (Ref. TABLE 3) in relation with to a set limit. The limit switch can be configured in one of the four mode given below:-

1) Hi alarm & Energized Relay

2) Hi alarm & De-Energized Relay

3) Lo alarm & Energized Relay

4) Lo alarm & De-Energized Relay

With User selectable Trip point, Hysteresis, Energizing Delay & De-Energizing delay.

Hi Alarm:

If Hi-Alarm Energized or Hi Alarm De-Energized option is selected then relay will get energized or De-energized, if selected parameter is greater than or equal to trip point.

Lo Alarm:

If Lo-Alarm Energized or Lo Alarm De-Energized option is selected then relay will get energized or De-energized, if selected parameter is less than or equal to trip point.

Note: For Lo-Alarm configuration, set the values of trip point & hysteresis such that % trip point + % hysteresis should be less than 100% Value.

Example for Phase angle:

If trip point is set 70% then maximum applicable hysteresis is 42.8%. i.e Trip point 70% (252°) + Hysteresis 42.8% (107.8°) = 359.8° If total value is greater than the 100% i.e. 360° then relay will not release.

Example for PF:

For Hi-Alarm Energized, if trip point is 70% & hysterisis is 30%, then trip value = $0.7\times90^\circ$ = 63° . Tripping PF = $\cos(63)$ =0.4539 & hysterisis = 0.3×0.4539 = 0.136. Hence, the relay will energize above 0.4539 and deengrize below 0.3179.

Note: This function will work irrespective of +/sign. It depends only on value.



Trip point:

Trip point can be set in the range as specified in TABLE 3 of nominal value for Hi-Alarm & 10% to 100% of nominal value for Lo-Alarm.

Hysteresis:

Hysteresis can be set in the range of 0.5% to 50 % of set trip point. If Hi-alarm Energized or Hi-alarm De-energized is selected then relay will get De-energized or Energized respectively, if set parameter value is less than Hysteresis. Similarly if Lo-alarm Energized or Lo-alarm De-Energized.

Note : In case of Io alarm if trip point is set greater than 80% then the maximum hysteresis can be set such that the total Trip point+ Hysteresis(% of trip point value) will not exceed 120% of range.

For example : If trip point is set at 90%, then maximum 33.3% hysteresis should be set such that, [90 + 29.99 (33.3% of 90)] = 120.

Energizing Delay:

The energizing delay can be set in the range from 1 to 9999 seconds.

De-Energizing Delay:

The De-energizing delay can be set in the range from 1 to 9999 seconds.

Examples of different configurations

Parameter No. 4 (Current1) Trip Point = 50% Hysteresis = 50% of trip point Energising Delay: 2 sec De-energising Delay: 2 sec



Parameter No.	Parameter	3P 4W	3P 3W	1P 2W	Trip Point Set Range	100% Value
0	None	√	\checkmark	\checkmark	-	-
1	Volts 1	√	✓	✓	10 - 120 %	Vnom (L-N)
2	Volts 2	√	✓	×	10 - 120 %	Vnom (L-N)
3	Volts 3	✓	✓	×	10 - 120 %	Vnom (L-N)
4	IL1	✓	\checkmark	\checkmark	10 - 120 %	Inom
5	IL2	✓	\checkmark	×	10 - 120 %	Inom
6	IL3	✓	✓	×	10 - 120 %	Inom
7	W1	✓	×	\checkmark	10 - 120 %	Nom (3)
8	W2	✓	×	×	10 - 120 %	Nom (3)
9	W3	✓	×	×	10 - 120 %	Nom (3)
10	Va1	√	×	\checkmark	10 - 120 %	Nom (3)
11	Va2	✓	×	×	10 - 120 %	Nom (3)
12	Va3	√	×	×	10 - 120 %	Nom (3)
13	Var1	✓	×	\checkmark	10 - 120 %	Nom (3)
14	Var2	✓	×	×	10 - 120 %	Nom (3)
15	VAr3	√	×	×	10 - 120 %	Nom (3)
16	Volts Ave.	✓	✓	×	10 - 120 %	Vnom (2)
17	Current Ave.	✓	✓	×	10 - 120 %	Inom
18	Watts sum	✓	\checkmark	×	10 - 120 %	Nom (3)
19	VA sum	✓	\checkmark	×	10 - 120 %	Nom (3)
20	VAr sum	✓	✓	×	10 - 120 %	Nom (3)
21	PF Ave. #	✓	✓	×	10 - 90 %	90°
22	PA Ave. #	✓	\checkmark	×	10 - 90 %	360°
24	Freq. #	√	✓	√	10 - 90 %	66 Hz (1)
27	Wh Import	✓	\checkmark	√	10 - 9999999	Nom (3)
29	Wh Export	✓	✓	✓	10 - 9999999	Nom (3)
31	VAr Capacitive	\checkmark	\checkmark	\checkmark	10 - 9999999	Nom (3)
32	VAr Inductive	\checkmark	\checkmark	\checkmark	10 - 9999999	Nom (3)
34	VA	\checkmark	\checkmark	\checkmark	10 - 9999999	Nom (3)
36	VL1-L2	\checkmark	×	×	10 - 120 %	Vnom (L-L)
37	VL2-L3	\checkmark	x	×	10 - 120 %	Vnom (L-L)

TABLE 3 : Parameters for Limit output

Parameter No.	Parameter	3P 4W	3P 3W	1P 2W	Trip Point Set Range	100% Value
38	VL3-L1	√	×	×	10 - 120 %	Vnom (L-L)
39	I Neutral	✓	×	×	10 - 120 %	Inom
40	Relay Manual OFF	✓	<	√	1	-
41	Relay Manual ON	✓	<	√	1	-

TABLE 3 : Continued...

Note : Parameters 1,2,3 are L-N Voltage for 3P 4W & L-L Voltage for 3P 3W.

#Note : Refer #Note of Section 5.2 for details.

(1) For Frequency 0% corresponds to 45 Hz and 100% corresponds to 66 Hz.

(2) For 3P 4W and 1P2W the nominal value is VLN and that for 3P 3W is VLL.

(3) Nominal Value for power is calculated from Nominal Voltage and current values.

(4) Nominal Value is to be considered with set CT/ PT Primary values.

(5) For single phase L1 Phase values are to be considered as System values.

5.3 Timer Output

Timer output can be used to operate the Relay in a cyclic manner. The user can define the ON period and OFF period and also the number of times this cycle is to be repeated. The number of Cycles (N) can be indefinite or 1 to 9999. The counting is shown on a measurement screen as explained before.



TABL	.E	5	:	Datalogging	Parameters	List
------	----	---	---	-------------	------------	------

Para. No.	Parameter	3P 4W	3P 3W	1P 2W
0	V1	✓	✓	✓
1	V2	✓	\checkmark	×
2	V3	\checkmark	\checkmark	×
3	11	\checkmark	\checkmark	\checkmark
4	12	\checkmark	\checkmark	×
5	13	\checkmark	\checkmark	×
6	W1	\checkmark	×	\checkmark
7	W2	\checkmark	×	×
8	W3	\checkmark	×	×
9	VA1	✓	×	\checkmark
10	VA2	✓	×	×
11	VA3	✓	×	×
12	VAR1	✓	×	\checkmark
13	VAR2	✓	×	×
14	VAR3	✓	×	×
15	PF1	×	×	×
16	PF2	×	×	×

Para. No.	Parameter	3P 4W	3P 3W	1P 2W
17	PF3	×	×	×
18	Angle1	x	×	×
19	Angle2	×	×	×
20	Angle3	x	×	×
21	Volt Avg	✓	✓	×
22	Volt Sum	✓	✓	×
23	Current Avg	✓	✓	×
24	Current Sum	✓	✓	×
25	Watt Avg	\checkmark	\checkmark	×
26	Watt Sum	\checkmark	✓	×
27	VAAvg	\checkmark	✓	×
28	VA Sum	✓	✓	×
29	VAR Avg	✓	✓	×
30	VAR Sum	✓	✓	×
31	PF Avg	✓	✓	×
32	PF Sum	×	×	×
33	Phase Angle Avg	 ✓ 	✓	×
34	Phase Angle Sum	x	×	×
35	Freq	✓	✓	\checkmark
36	Wh import	✓	✓	\checkmark
37	Wh export	✓	✓	✓
38	VARh Capacitive	✓	✓	✓
39	VARh Inductive	✓	✓	√
40	VAh	\checkmark	\checkmark	\checkmark
42	kw imp demand	×	×	×
43	max kW imp demand	x	×	x
44	kW exp demand	x	×	x
45	max kW exp demand	×	×	×
46	kVAr Cap. demand	x	×	x

TABLE 5: Continued...

Para. No.	Parameter	3P 4W	3P 3W	1P 2W
47	max kVAr Cap. demand	x	x	x
48	kVAr Ind. demand	×	×	×
49	max kVAr Ind. demand	×	×	×
50	KVA demand	×	x	×
51	max KVA demand	×	×	×
52	current demand	×	×	×
53	max current demand	×	x	×
54	Wh Import Overflow Count	✓	✓	√
56	Wh Export Overflow Count	✓	√	✓
58	VARh Capacitive OF Count	✓	✓	✓
60	VARh Inductive OF Count	✓	✓	✓
62	Vah Overflow Count	✓	✓	✓
66	system voltage max	✓	✓	✓
67	system voltage min	✓	√	✓
68	RPM	✓	✓	√
70	system current max	✓	√	✓
71	system current min	✓	✓	✓
100	V12	✓	×	×
101	V23	✓	×	×
102	V31	✓	×	×
103	V THD-L1	×	×	×
104	V THD-L2	×	×	×
105	V THD-L3	×	×	×
106	I THD-L1	x	×	×
107	I THD-L2	×	×	×
108	I THD-L3	x	×	×
109	System V-THD	✓	✓	\checkmark
110	System I-THD	✓	✓	√
112	Neutral Current	✓	×	×

TABLE 5: Continued...

Para. No.	Parameter	3P 4W	3P 3W	1P 2W
113	Run hour	✓	✓	✓
114	On Hour	✓	✓	✓
115	No. of interrupts	✓	✓	✓
166	Phase indicate	 ✓ 	✓	×

TABLE 5: Continued...

6. Phasor Diagram :

Quadrant 1: 0° to 90° Quadrant 3: 180° to 270° Quadrant 2: 90° to 180° Quadrant 4: 270° to 360°



Connections	Quadrant	Sign of Active Power (P)	Sign of Reactive Power (Q)	Sign of Power Factor (PF)	Inductive / Capacitive
Import	1	+ P	+ Q	+	L
Import	4	+ P	- Q	+	С
Export	2	- P	+ Q	-	С
Export	3	- P	- Q	-	L

Inductive means Current lags Voltage

Capacitive means Current leads Voltage

When Multifunction Meter displays Active power (P) with " + " (positive sign), the connection is "**Import** ". When Multifunction Meter displays Active power (P) with " - " (negative sign), the connection is "**Export**".

7. Installation

Mounting of the Meter is featured with easy "Clipin" mounting. Push the meter in panel slot (size 92 x92 mn), it will click fit into panel with the four integral retention clips on two sides of meter. If required, additional support is provided with swivel screws as shown in figure.

The front of the enclosure conforms to IP54. Additional protection to the panel may be obtained by the use of an Optional panel gasket. The terminals at the rear of the product should be protected from liquids.

The Meter should be mounted in a reasonably stable ambient temperature and where the operating temperature is within the range 0 to $50^\circ C$. Vibration should be kept to a minimum and the product should not be mounted where it will be subjected to excessive direct sunlight.



Caution

1. In the interest of safety and functionality this product must be installed by a qualified engineer, abiding by any local regulations.

Voltages dangerous to human life are present at some of the terminal connections of this unit. Ensure that all supplies are deenergised before attempting any connection or disconnection.

3. These products do not have internal fuses therefore external fuses must be used to ensure safety under fault conditions.

7.1 EMC Installation Requirements

This product has been designed to meet the certification of the EU directives when installed to a good code of practice for EMC in industrial environments, e.g.

1. Screened output and low signal input leads or have provision for fitting RF suppression components, such as ferrite absorbers, line filters etc., in the event that RF fields cause problems.

Note : It is good practice to install sensitive electronic instruments that are performing critical functions, in EMC enclosures that protect against electrical interference which could cause a disturbance in function.

2. Avoid routing leads alongside cables and products that are, or could be, a source of interference.

3. To protect the product against permanent damage, surge transients must be limited to 2kV pk. It is good EMC practice to suppress differential surges to 2kV at the source. Theunit has been designed to automatically recover in the event of a high level of transients. In extreme circumstances it may be necessary to temporarily disconnect the auxiliary supply for a period of greater than 5 seconds to restore correct operation. The Current inputs of these products are designed for connection in to systems via Current Transformers only, where one side is grounded.

4. ESD precautions must be taken at all times when handling this product.



7.2 Case Dimensions and Panel Cut-Out

7.3 Wiring

Input connections are made directly to screw-type terminals with indirect wire pressure. Numbering is clearly marked on the connector. Choice of cable should meet local regulations. Terminal for both Current and Voltage inputs will accept upto 4mm² (12AWG) solid or 2.5 mm² stranded cable. Note: It is recommended to use wire with lug for connection with meter.

7.4 Auxiliary Supply

Meter should ideally be powered from a dedicated supply, however powered from the signal source, provided the source remains within it may be the limits of the Chosen auxiliary voltage range.

7.5 Fusing

It is recommended that all voltage lines are fitted with 1 Amp HRC fuse.

7.6 Earth/Ground Connections

For safety reasons, CT secondary connections should be grounded in accordance with local regulations.

8. Connection Diagrams





3-PHASE 4-WIRE UNBALANCED LOAD DIGITAL METERING SYSTEM (WITH EXTERNAL AUX.)

SINGLE PHASE 2-WIRE DIGITAL METERING SYSTEM (WITH EXTERNAL AUX.) (୨) ମୋ(ମାର) ଶିରାହା

> AUX SUPPLY

> > Ă

9. Pluggable Module



Auxiliary Supply

External Higher Aux. External Higher Aux. Nominal Value Aux Frequency Range VA Burden With Addon card VA Burden With Ethernet card

Operating Measuring Ranges

Voltage Current

Starting Current

Frequency Power Factor Total Harmonic Distortion

Reference conditions for Accuracy

Reference temperature (as per IEC 62053-21)

Accuracy

Voltage

Current

Frequency

Active power

Reactive power

100V to 550V AC-DC 230V AC/DC 50/60 Hz for AC Aux 45 to 65 Hz < 6 VA approx. (at nominal) < 8 VA approx. (at nominal)

20 ... 120 % of nominal Value 1....120% of nominal value

as per IEC 62053-22 (0.5s/0.2s)

45 to 66 Hz 0.5 Lag ... 1 ... 0.8 Lead 50% upto 15th harmonic 10% upto 31st harmonic

23°C <u>+</u> 2°C

- \pm 0.5% of nominal value \pm 0.5% of nominal value \pm 0.2 % of mid frequency \pm 0.5% of nominal value
- ± 1.0% of nominal value

10. Specification

System

3 Phase 3 Wire / 4 Wire or Single Phase programmable on site

Inputs

Nominal Input Voltage (AC RMS) System PT Primary Values System PT Secondary Values Max continuous input voltage Nominal input voltage burden Nominal Input Current max continuous input current Nominal input current burden System CT primary values System CT secondary values Overload Indication Overload withstand

Voltage input

Current input

100VLL to 600VLL programmable on site. (57.7VLN to 346.4VLN) 100VLL to 1200 kVLL, programmable on site 100VLL to 600VLL programmable on site. (57.7VLN to 346.4VLN) 120% of Nominal Value <0.3 VA approx, per Phase (at nominal 240V) 1A/5A 120% of Nominal value <0.3 VA approx. per phase Std. Values 1 to 9999A (1 or 5 Amp secondary) 1A / 5A, programmable on site "-OI -" >121% of Nominal value (for voltage and current) 2 x Rated Value

(1s application at 10s intervals) repeated 10 times
 2 x nominal Value
 (1s application at 5min intervals) repeated 5 times

Apparent Power	<u>+</u> 0.5% of nominal value
Power Factor / Phase Angle	<u>+</u> 3°
Active energy	as per IEC 62053-22 (0.5s) as per IEC 62053-22 (0.2s) (optional)
Reactive energy	Class 2 as per IEC 62053-23
Apparent energy	class 1
THD (Voltage / Current)	<u>+</u> 3.0 %

Accuracy (for optional 0.2s)

± 0.2% of nominal value
± 0.2% of nominal value
± 0.2 % of mid frequency
± 0.2% of nominal value
± 0.2% of nominal value

Display

LCD Display with backlit Response time to step input

Controls User Interface

Standards EMC Immunity Immunity

Safety

IP for water & dust

Approx. 1 sec.

3 push buttons

IEC 61326 - 1 : 2012 IEC 61000-4-3. 10V/m -Level 3 industrial Low level IEC 61010-1-2010, permanently connected use IEC 60529

Pollution degree	2
Installation Category	III
Isolation	
Protective Class	2
High Voltage Test	
Input+Aux vs Surface	4 kV RMS, 50Hz, 1 min
Input vs Remaining Circuit	3.3 kV RMS, 50Hz, 1 min
Environmental conditions	
Operating temperature	-10 to +60 °C

Storage temperature Relative humidity

Warm up time

Shock (As per

Vibration

Enclosure front

Dimensions Bezel Size

Panel cut out

Overall Depth

Weight

Panel thickness

IEC 60068-2-27)

Number of Sweep cycles

Enclosure back (Terminals)

-10 to +60 °C -20 to +70 °C 0 .. 95 % RH (Non condensing) 3 minute (minimum) Half Sine wave, Peak acceleration 30gn (300 m/s^2), duration 18 ms 10..150..10 Hz, 0.15mm amplitude 10 per axis IP 54 IP 20

96mm x 96mm DIN 43718 92mm+0.8 X 92mm+0.8 75 mm 1 - 3mm for self clicking 1 - 6mm for swivel screws 320 grams Approx.

Pulse output Option	
Relay	1NO
Switching Voltage & Current	240 VAC , 5 A.
Default Pulse rate Divisor	1 per Wh (up to 3600W), 1 per kWh (up to 3600kW), 1 per MWh (above 3600 kW up to 30000 kW)
Pulse rate Divisors	Programmable on site
10	1 per 10Wh (up to 3600W), 1 per 10kWh (up to 3600kW), 1 per 10MWh (above 3600 kW up to30000 kW)
100	1 per 100Wh (up to 3600W), 1 per 100kWh (up to 3600 kW), 1 per 100MWh (above 3600 kW up to 30000 kW)
1000	1 per 1000Wh (up to 3600W), 1 per 1000kWh (up to 3600kW), 1 per 1000MWh (above 3600 kW up to 30000 kW)
Pulse Duration	60ms , 100ms or 200ms
Note : 1. Refer TABLE 4 for details. 2. Above conditions are also applicab ModBus (RS 485) Option :	le for Reactive & Apparent Energy.
Protocol	ModBus (RS 485)
Baud Rate Parity	4.8k, 9.6k, 19.2k, 38.4k, 57.6kbps Odd or Even, with 1 stop bit, Or None with 1 or 2 stop bits
Ethernet	Ethernet access on Modbus TCP/IP Protocol
Impulse Output : Impulse LED	For Energy testing

Impulse Constant

Depending on nominal system nominal power, the number of impulses are created to measure the energy. The number of impulses for particular nominal power is set which indicates 1kWh energy. Energy can be Watt, VA or Var. Following table shows impulses corresponding to nominal system energy.

System nominal power	Impulse constant
<=400	16000
<=800	8000
<=1600	4000
<=3200	2000
>3200	1000

11. Connection for Optional Pulse Output / RS 485 (rear view of Multifunction Meter):

1. RS 485 Output with Relay1 & Relay2



12. Ordering codes

Model	Key features
END25RS	RS485 interface + 2 Pulses outputs, accuracy class 0.5s
END25ETH	Ethernet + data logging, accuracy class 0.2s



ETI Elektroelement d.o.o. 1411 Izlake Slovenia Tel. : +386 03 56 57 570 e-mail: eti@eti.si